# Center for Economic Institutions Working Paper Series

No. 2009-9

"The Effects of Focus Versus Diversification on Bank Performance: Evidence from Chinese Banks"

> Allen N. Berger, Iftekhar Hasan and Mingming Zhou

> > November 2009



Institute of Economic Research Hitotsubashi University 2-1 Naka, Kunitachi, Tokyo, 186-8603 JAPAN <u>http://cei.ier.hit-u.ac.jp/English/index.html</u> Tel:+81-42-580-8405/Fax:+81-42-580-8333

### The effects of focus versus diversification on bank performance: Evidence from Chinese banks

Allen N. Berger<sup>a</sup>, Iftekhar Hasan<sup>b,\*</sup>, Mingming Zhou<sup>c</sup>

<sup>a</sup>University of South Carolina, Columbia, SC 29208 U.S.A. <sup>a</sup>Wharton Financial Institutions Center, Philadelphia, PA 19104 U.S.A. <sup>a</sup>CentER, Tilburg University, The Netherlands <sup>b</sup>Lally School of Management & Technology, Rensselaer Polytechnic Institute, Troy, NY 12180 U.S.A. <sup>b</sup>Bank of Finland, Helsinki, Finland <sup>c</sup>College of Business and Administration, University of Colorado at Colorado Springs, Colorado Springs, CO 80918 U.S.A.

#### Abstract

This paper investigates the effects of focus versus diversification on bank performance using data on Chinese banks during the 1996-2006 period. We construct a new measure, *economies of diversification*, and compare the results to those of the more conventional *focus index*, which is based on the sum of squares of shares in different products or regions. Diversification is captured in four dimensions: loans, deposits, assets, and geography. We find that all four dimensions of diversification are associated with reduced profits and higher costs. These results are robust regardless of alternative measures of diversification and performance. Furthermore, we observe that banks with foreign ownership (both majority and minority ownership) and banks with conglomerate affiliation – are associated with fewer diseconomies of diversification, suggesting that foreign ownership and conglomerate affiliation play an important mitigating role. This analysis may provide important implications for bank managers and regulators in China as well as in other emerging economies.

*JEL classification:* G21; G28; G34 *Keywords:* Diversification; Focus; Efficiency; Chinese Banking

The authors thank the anonymous referees and the participants in the Conference on Performance Measurement in the Financial Services Sector: Frontier Efficiency Methodologies and Other Innovative Techniques for helpful comments.

<sup>\*</sup> Corresponding author. Tel: +1 518 276 2525; Fax: +1 518 276 8661; <u>hasan@rpi.edu</u> (I. Hasan).

#### 1. Introduction

Should banks diversify across different products and geographic regions, or should they specialize? The focus versus diversification literature is well established in corporate finance, although there is not a general consensus as to whether conglomerates tend to perform better or more poorly than focused firms. Moreover, the findings in the general corporate finance literature may or may not apply to the banking sector, because banks are different from other firms.<sup>1</sup> Sometimes banks also face conflicting regulation and supervision that create incentives to either focus or diversify. Branching, entry, and asset investment restrictions often encourage focus, while supervisors tend to encourage diversification to reduce risks.

Although there are some studies on the link between diversification and performance of banks, there is no consensus thus far, with evidence supporting both arguments. Proponents of diversification suggest that diversified banks can benefit from leveraging managerial skills and abilities across products and geographic regions (Iskandar-Datta and McLaughlin (2005)), gaining economies of scope through spreading fixed costs over products and regions (Drucker and Puri (2009)), and providing a financial supermarket to customers who demand multiple products.<sup>2</sup> On the other side, proponents of focus argue that diversified banks can suffer from diluting the comparative advantage of management by going beyond their existing expertise (Klein and Saidenberg (1998)), diversification-inducing competition (Winton (1999)), and increased agency costs resulting from value-decreasing activities of

<sup>&</sup>lt;sup>1</sup> Important differences between banks and firms in other industries are that banks are delegated monitors (Diamond, 1984) and proprietary information acquirers of the borrowers (Fama (1980, 1985), James (1987), Sharpe (1990), Rajan (1992)), and banks, by their very nature, are designed to diversify (Winton (1999), Acharya et al. (2006)).

<sup>&</sup>lt;sup>2</sup> Diversified financial institutions may also reduce the expected costs of financial distress or bankruptcy by lowering risks through spreading operations across different products or economic environments (Boot and Schmeits (2000)). In addition, geographically diversified firms may obtain tax benefits by transferring income from high tax areas to low tax areas (Iskandar-Datta and McLaughlin (2005)).

the managers who have lowered their personal risk (Amihud and Lev (1981), Deng and Elyasiani (2007), Laeven and Levine (2007)).<sup>3</sup>

Besides the inconclusive findings in the literature, the empirical evidence documented on banking diversification to date is primarily based on the U.S. market and other developed countries, with much less insight and discussion on the banking industry in emerging or transitional economies (Odesanmi and Wolfe (2007) may be one of the few exceptions). When considering the size and impact of some emerging markets such as China on the world economy, one might be surprised to notice that there is a big gap in the banking literature: there are no empirical studies documenting the effects of diversification strategies on performance of Chinese banks.

China boasts one of the biggest and fastest-growing emerging economies in the world, with an average of about 10% GDP growth per year in real terms over the last two decades, and is projected by some to become the world's largest economy in the coming decades. Chinese banks have been playing a big role in channeling the financial resources between the savings of households, government deposits and transfers, and the financing of the Chinese enterprises, as the Chinese stock market did not exist until 20 years ago and has been only serving limited number of companies which are favored by the government. At the same time, however, Chinese banks have been heavily influenced by policy makers. For example, although most of the national banks in China enjoy more freedom to spread their businesses and services across the nation, most of the regional and city commercial banks face strict geographical restrictions, and throughout the last few decades, the commercial banks have faced limited degrees of freedom in terms of the products and services they can offer.<sup>4</sup> But these strict regulations and limitations are becoming more relaxed, especially since China became a member of the

<sup>&</sup>lt;sup>3</sup> Furthermore, diversification of banking activities across international borders can lead to increased political risk, foreign exchange risk, and difficulties of dealing with different languages, laws, and cultures, which can destroy shareholder value (Miller and Parkhe (2002), Fauver et al. (2004), Deng and Elyasiani (2007)).

<sup>&</sup>lt;sup>4</sup> For example, the People's Republic of China (PROC) issued 'Regulations on Credit Cards Business' (effective on April 1, 1996), which applies to commercial banks, institutions and individuals which hold, use, or accept credit cards in China. In Chapter I of this regulation, the fifth clause says "Commercial banks, without the permission from PROC, cannot issue credit cards..."

World Trade Organization (WTO) in 2001. Given the changing regulations and their relative new experience to the business freedom and different options they have, banks in China may not necessarily know their optimal product and geographical strategies. This paper therefore brings evidence on the potential benefits of being focused versus diversified.

Based on the economies of diversification framework, we measure the differences in predicted profits and costs between diversified and hypothetical focused banks. Diversification is captured in four dimensions: loans, deposits, assets, and geography. We find that all the four dimensions of diversification decrease profits and increase costs for the Chinese banks after controlling for risks, and these results maintain regardless of different alternative measures of diversification and performance. Additionally, we find that foreign ownership and conglomerate affiliation play an important mitigating role in the diversification discount of the Chinese banks, in the sense that the banks with foreign ownership (either majority foreign or minority foreign) or conglomerate affiliation experience fewer diseconomies of diversification, i.e., suffer less loss of profits and less increase in costs when they diversify.

This paper contributes to the existing literature in the following ways. First, we provide a new measure of economies of diversification and apply it to the banking sector. Second, our paper fills the gap in the existing banking diversification-performance literature which is heavily focused on the developed markets. By presenting and discussing evidence of the diversification premium/discount in the Chinese banking industry, our paper provides important insights into one of the largest emerging and transitional economies. In this regard, this paper should not only provide practical implications for Chinese bank managers, but also lend some perspectives to the policy makers in China and other emerging economies as well, who set rules that encourage and/or discourage the diversification of banking. Finally, our paper presents evidence on the effects of foreign bank ownership and

conglomerate affiliation on the benefits/costs of diversification, which may aid in the discussion of policies regarding foreign bank entry.

The remainder of this paper is organized as follows. Section 2 reviews some of the research literature on the effect of banking diversification on the performance. Section 3 gives background information on the Chinese banking industry. Section 4 shows our data on the Chinese banks and outlines our empirical methodology. Section 5 displays our empirical results, and Section 6 concludes.

#### 2. Literature review

The issue of focus versus diversification is well documented in the corporate finance literature, although a general consensus has not been achieved. For example, while many studies provide evidence that conglomerates perform more poorly than specialized firms (e.g., Lang and Stulz (1994), Berger and Ofek (1995), Lamont and Polk (2002)), other studies support the opposite conclusion. For example, Villalonga (2004) uses the Business Information Tracking Series (BITS) and finds diversification premium for a sample that yields a discount based on the segment data used in previous studies, which supports the argument that the diversification discount found by earlier studies is an artifact of segment data. Some international studies also present heterogeneities of diversification effects on performance across different countries and institutional settings. For example, Khanna and Yafeh (2005) examine the hypothesis that business groups facilitate mutual insurance among affiliated firms and find substantial evidence of risk sharing by Japanese, Korean, and Thai groups, but little evidence of it elsewhere in the world. More recently, Khanna and Yafeh (2007) present evidence about reasons for formation, prevalence, and evolution of diversified business groups in different environments, and they argue that business groups are responses to different economic conditions and that, from a welfare standpoint, they can sometimes be "paragons" and, at other times, "parasites."

However, the results in the general literature may or may not apply to banks, because banks differ from other industries as discussed above.

The existing banking literature does not provide consensus as to the question "should banks diversify portfolios and geographic regions, or should they specialize?" Instead, there is evidence supporting both arguments. Traditional arguments suggest that banks should be as diversified as possible, as banks are typically highly levered, and diversification across sectors reduces their chance of costly financial distress/bankruptcy. Several models of intermediation suggest that diversification makes it cheaper for institutions to achieve credibility in their role as screeners or monitors of borrowers (see Diamond (1984), Ramakrishnan and Thakor (1984), and Boyd and Prescott (1986)). On the other hand, some studies (see Jensen (1986), Berger and Ofek (1996), Servaes (1996), and Denis et al. (1997)) argue and/or provide evidence that financial institutions should focus on a single line of business so as to take greatest advantage of management's expertise and reduce agency problems, leaving investors to diversify on their own.

Cases of bank crises/failures also lend support to both sides. Proponents of a diversification premium can cite Continental Illinois' failure in 1984, linked to a large concentration of energy-sector loans, and Bank of New England's failure over 1989 - 1991, linked to a large concentration in New England commercial real estate loans. Proponents of diversification discount can cite the credit problems that followed rapid diversification at many institutions during the 1980s, including Citicorp, Bank of America, Credit Lyonnais, and a number of major Japanese banks; the current credit crisis in which banks expanded into securities created by other institutions with risks they did not understand; a number of recent cases in which focus has produced superior results without excessive risk, including Comerica, a specialist in middle market and small business lending, and FirstUSA and MBNA, which are specialists in credit card lending.

Further, there are some studies that argue that expected costs of financial distress or bankruptcy may be reduced by spreading operations across different economic environments (see Boot and Schmeits (2000)). For example, Boyd and Graham (1988) study hypothetical mergers between banks and securities firms, real estate companies, and insurance firms, and they find that banks merging with insurance companies might reduce the risk of bankruptcy, while mergers with securities/and real estate companies would increase the risk of bankruptcy.<sup>5</sup> Rose (1989) suggests that banks moving into non-bank product lines could reduce cash flow risk. Templeton and Serveriens (1992) find that banks diversifying into other financial services would reduce unsystematic risk, while there was no effect on systematic risk. Berger et al. (1999) find that consolidation in the financial services industry has been consistent with greater diversification of risks on average but with little or no cost efficiency improvements.

However, more recent studies suggest that costs may outweigh benefits when banks choose to diversify their product offerings. For example, DeYoung and Roland (2001) find that U.S. banks replacing traditional lending activities with fee-based activities are associated with higher revenue volatility, implying higher earnings volatility (risk).<sup>6</sup> Similarly, Stiroh (2004) documents that non-interest diversification is negatively linked with performance. Acharya et al. (2006) provide results suggesting that there are diseconomies of scope that arise through weakened monitoring incentives and a poorer-quality loan portfolio when a risky bank expands into additional industries and sectors, complementing the agency-theoretic analysis of the boundaries of a bank's activities as proposed in Cerasi and Daltung (2000). Laeven and Levine (2007) find that financial conglomerates engaging in multiple lending activities have lower market values than they would if they were broken into separate

<sup>&</sup>lt;sup>5</sup> Boyd and Graham (1988) studies hypothetical mergers because at that time, the activity-based diversification was not yet permitted in the U.S. This study was updated by Lown et al. (2000) with similar findings.

<sup>&</sup>lt;sup>6</sup> Demsetz and Strahan (1997) also found that large BHCs were better diversified than small BHCs, but this diversification did not translate into lower risk levels because of lower capital ratios, larger commercial and industrial loan portfolios, and greater use of derivatives by large banks.

financial institutions, and their results are consistent with theories that stress intensified agency problems in financial conglomerates that engage in multiple activities and indicate that any economies of scope are not sufficiently large to produce a diversification premium.

Besides the diversification in product and line of service dimensions, there is also a trend toward banks diversifying geographically. Banks have the potential to achieve economies of scale in geographical dimension, because once an initial investment is made and the basic infrastructure is in place, organizations can expand the system elsewhere at a potentially reduced cost. Benefits of geographical diversification include: better access to capital markets in other regions/countries, which potentially leads to reduced cost of capital (see Deng and Elyasiani (2007)), greater market power (see Iskandar-Datta and McLaughlin (2005)), and reduced tax liabilities as geographically diversified banks can transfer resources from high tax areas to low tax areas. Consistent with these arguments, Mahajan et al. (1996), based on U.S. multinational and domestic banks from 1987-1990, conclude that multinational banks were able to fully exploit economies of scale, and had lower inefficiencies than domestic banks.

On the other hand, some studies argue that there are costs associated with geographic diversification. First, the classical analysis of agency costs (e.g., Amihud and Lev (1981)) can also be applied to diversified banks, where diversification may help reduce the personal risk for the manager instead of creating value for the shareholders. The findings of Deng and Elyasiani (2007) support this line of argument. Meanwhile, some other disadvantages associated with global diversification, such as increased exchange rate and political risk and difficulties of dealing with different languages, laws and customs can also destroy shareholder's value (e.g., Miller and Parkhe (2002), Fauver et al. (2004), Deng and Elyasiani (2007)). Berger et al. (2000a) examine the efficiency of cross-border consolidations of financial institutions from France, Germany, Spain, the U.K., and the U.S., and find that domestic banks have higher profit efficiency than do foreign banks, except for US-based foreign

banks. DeLong (2001) examines U.S. bank mergers with respect to both activity and geographic location and find that banks focusing on both activity and geography were value increasing. Finally, some studies also find that geographic diversification can also lead to organizational inefficiency (e.g., Klein and Saidenberg (1998)).

A few studies use the concepts of revenue and profit scope economies in the banking literature. For example, Berger et al. (1993) analyze profit scope economies using the standard profit function. Berger et al. (1996) analyze revenue scope economies using the alternative revenue function, and Clark and Siems (1997) use the alternative profit function to evaluate expansion-path scale economies. These studies generally do not find consistent benefits of either joint production or specialization within banking industry. Moreover, most of the studies on cost scope economies within financial service industries find no substantial evidence of cost scope economies (e.g., Berger et al. (1987), Mester (1987, 1993), Hunter et al. (1990), Berger and Humphrey (1991), Goldberg et al. (1991), Pulley and Humphrey (1993), Noulas et al. (1990)).

Importantly, the existing banking literature on focus – diversification issue is heavily concentrated in US and European banking markets, while leaving this issue in emerging/transitional economies largely unexamined. Our paper attempts to fill this gap in the literature by studying the economies of diversification in the largest emerging and transitional economy in the world, i.e., China.

#### 3. Chinese banks: their background and experience

In this section, we present a brief overview of the diversification experience of the Chinese banking system, with our main focus on the post-1990s period. For a more general and detailed description and discussion about the background of Chinese banking sector, see Berger et al. (2009).

Chinese banks in the past did not have much choice in terms of product diversification. They

were often required by policy makers to direct their loans to certain sectors or to certain customers (e.g., provide policy loans). Consider the Big Four banks as examples. Agricultural Bank of China (ABC) was required to direct majority of their loans to agricultural sector, especially in rural areas. China Construction Bank (CCB) was instructed to provide majority of their lending to real estate and construction materials industries. Industrial and Commercial Bank of China (ICBC) was mandated to lend heavily to manufacturing and commercial sectors. Bank of China (BOC) was advised to focus on the foreign exchange and currency businesses. Such restrictions gradually disappeared in the latter half of the 1990s.

In addition, Chinese regulators used to insist on strict guidelines separating commercial banking, investment banking (including asset management), insurance, and trust businesses, believing that restrictive firewalls between different financial services can prevent spillover of financial crises, as coded in 1995 Commercial Banking Law. It was not until recently that some of the restrictions began to loosen. For example, although the Chinese government is still reluctant to grant commercial banks licenses for investment banking business (such licensing became very unlikely since the breakout of financial crisis in US), more cooperation between commercial banks and insurance companies has recently been encouraged by regulators.<sup>7</sup> Meanwhile, the issuance of the 'Guidance of Cooperation between the Operations of Banks and Trust Companies' by China Banking Regulatory Commission (CBRC) in December 2008 seems to suggest that the Chinese regulators are more open to the idea of cooperation between nontraditional banking businesses and commercial bank operations.

Even within the commercial banking business, the Chinese banks historically rely heavily on net interest income, with the fee-based business being much more underdeveloped. According to Yuan (2006), most Chinese banks feature highly immature cash management and treasury businesses, with

<sup>&</sup>lt;sup>7</sup> On January 16, 2008, a Memorandum of Understanding on Strengthening Cooperation between the Banking and Insurance sectors and the Cross-sector Supervision was co-signed by the China Banking Regulatory Commission (CRBC) and the China Insurance Regulatory Commission, which provides a regulatory framework to guide to more formal cooperation between commercial banks and insurance companies.

fee-based business only accounting for about 10% of their total revenues on average. However, many Chinese banks are now focused on restructuring their operations and building their fee businesses, and newly emerging businesses such as cash management and wealth management show strong growth potential (Yuan (2006)). For example, Bank of China currently provides such fee-based business as paying fees, giving pay packets, handling stock fund transfer service, handling securities trading settlement, and handling insurance services.

Meanwhile, Chinese commercial banks also began to diversify into the trust business to seek broader growth. It is reported that Bank of Communications paid 1.22 billion RMB (US\$163 million) for 85 percent of Hubei International Trust & Investment Co., and the trust venture will provide services including asset packaging and financial consulting. China Minsheng Banking Corporation is also reported to have proposed to purchase a 25 percent stake in Shenzhen-listed Shaanxi International Trust for 2.34 billion RMB (US\$313 million).<sup>8</sup> This move is not surprising, as China conducted a revamp of its trust industry in early 2000s by closing down more than 200 trust companies, which borrowed heavily to invest in unprofitable projects. Since then, Chinese regulators have been hoping to nourish the sector by allowing firms to lure strategic investors. Chinese banks are also revving up bids to conduct business expansion by tapping fund management, private equity, and insurance industries to lessen their dependence on credit growth.

Universal banking practices were once the practice for some Chinese banks from 1985 to 1993, including some of the Big Four. However, the concept of universal banking was practically abandoned after the 1995 Commercial Bank Law became effective. To be aligned with WTO agreement, a new set of rules took effect to relax the restrictions on geographical restrictions on foreign banks operated in China, while restrictions on geographical diversification have been much more relaxed for the Chinese domestic banks. For example, all domestic banks (except the city commercial banks) are

<sup>&</sup>lt;sup>8</sup> Source: BoCom launches new trust to diversify growth prospects, Shanghai Daily (October 29, 2007).

permitted to establish branches and take deposits and lend across various regions in China. The most geographically diversified banks are the Big Four, and Bank of Communications (there is also a convention to term these five banks as "Big Five"), with each bank's branches extending to almost every corner of China, including both rural and urban areas. However, the great variety of economic and geographic conditions across regions in China contributes to the difficulty of information sharing, even within the same bank. Bank of China, for instance, with about 22,000 branches, only knows exactly how many branches it has six weeks after it closes its financial accounts. According to companies involved in the bank's ongoing technological overhaul, the state of the art of data transfer technologies nowadays cannot help as much when data from Tibet comes by camel train from the Gobi desert (see Robinson (2007)).

One of the interesting and unexplored areas of the Chinese banks' strategies is their expansion into foreign markets. In July 2007, China Development Bank took a small stake in Barclays Bank, helping to finance the British group's bid for a Dutch bank, ABN AMRO. This was the first time a Chinese bank took a stake in a European or American bank. The Chinese regulators have been very cautious in allowing big investments or takeovers by mainland banks in foreign ones, probably due to the concerns of the vast differences between the risk systems of the Chinese banks and the banks in the developed countries.

Overall, the environment of the Chinese banks is changing from little flexibility to more flexibility for the banks to choose between focused versus diversified strategies. With the forthcoming new rules and regulations under the WTO accession agreement, Chinese banks can choose a diversified or focused strategy.

#### 4. Data and Methodology

#### 4.1 Sample

Our sample is an unbalanced panel which includes financial data of 88 Chinese banks during the period 1996 to 2006, totaling 464 observations. All financial items are inflation-adjusted to the base year 1996. The basic data source is Bankscope - Fitch's International Bank Database. The major data source for branch data to construct geography diversification measures is annual issues of Almanac of China's Finance and Banking (ACFB), 1996-2006. Also, whenever Bankscope does not provide enough information or has questionable values, we collect or double-check the data from other official sources, such as annual issues of ACFB and the annual issues of China Statistical Yearbooks. Most of the sample banks follow Chinese Accounting Standards (CAS), while a few also prepared annual reports based on International Accounting Standards (IAS). The banks following IAS standards are primarily the joint venture banks (whose foreign ownership is more than 25% and less than 100%), foreign banks (100% owned by foreign), and banks listed in the domestic and/or foreign stock markets. While there are some differences between the CAS and IAS, it should be mentioned that the CAS was developed only in recent years following the principles of IAS. While we recognize that there may be some inconsistencies in financial data using different accounting standards, we do not find a material difference between the financial statements of the same bank while reporting under both CAS and IAS.

Among the 88 sample Chinese banks, there are the Big Four, which make up more than 66% of total commercial banking assets, and the 12 national shareholding commercial banks – known as the "second-tier" domestic banks – which own about 21% of commercial banking assets (based on ACFB statistics in 2006). Our sample also includes 50 city commercial banks and 22 joint venture/foreign banks, which make up about 4% of the total commercial banking assets in China in 2006.<sup>9</sup> In sum, our sample covers over 90% of the total assets of commercial banks in China.

For each bank in our sample, data are available to calculate the following portfolio

<sup>&</sup>lt;sup>9</sup> All the city commercial banks in China, along with the joint venture and foreign banks and other banking institutions, make up the "third-tier" banks in the industry, and they make up about 30% of the total banking assets in China (based on ACFB statistics in 2006).

#### decompositions:

- 1. A disaggregated loan decomposition based on each bank's industry loans, commercial loans, real estate loans, agriculture loans, and consumer loans.
- A disaggregated deposit decomposition based on each bank's customer demand deposits, customer savings deposits, non-bank corporate deposits, deposits by other banks, and other deposits.
- 3. A disaggregated asset decomposition based on each bank's total loans, deposits in other banks, financial investments (including investments in government securities, trading securities, and other financial assets), total fixed assets, and other assets.
- 4. A disaggregated geographical decomposition based on each bank's loans on various regions in China, including Northern, Eastern, Central, Southern, and Western regions. We follow the geographical definitions of the Chinese Statistics Bureau to classify the 31 provinces and municipalities into these five regions.

#### 4.2 *Economies of diversification approach*

In the following, we briefly discuss the approaches to measure economies of diversification and estimation issues, including functional form and variable selection. The section concludes by outlining a regression analysis of economies of diversification designed to determine the types of firms most likely to realize economies of diversification.

Because our approach to measure economies of diversification is based on the framework of economies of scope, we begin our discussion with scope economies. Profit scope economies are defined as the proportional increase in profits from producing given outputs by joint producers versus specialist firms. Likewise, cost scope economies are defined as the proportional increase in costs from producing given outputs by specializing firms versus joint producers.

It is important to choose an appropriate functional form for the profit (cost) function, so that the profits (costs) of a hypothetical focused firm can be predicted and then compared to the profits (costs) of the observed diversified firm to calculate scope economies. Once a specific functional form is adopted, the parameters of the profit (cost) function are estimated based on the observations of joint producers, with the assumption that the same functional form and parameters apply to the specialists as well. This is usually necessitated by an absence of data on specializing firms. For the selection of the profit (cost) function, we follow Berger et al. (2000b) and adopt a modified version of the composite form.<sup>10</sup> The composite functional form was first developed by Pulley and Braunstein (1992), and several studies have applied this form to study bank cost scope economies (Pulley and Humphrey (1993), McKillop, et al. (1996)) and bank revenue scope economies (Berger et al. (1996)).

The modified version of composite form that we adopt combines a quadratic structure for outputs and fixed netputs with a log-quadratic component for input prices, with interaction terms so that zero values for outputs are allowed, and separability is not imposed. Assume we estimate the measure the profit function for loan portfolios. We specify:

$$\frac{\pi}{z \times w_m} = \left[ \sum_{i=1}^n \beta_i \frac{L_i}{z} + \frac{1}{2} \sum_{i=1}^n \sum_{j=1}^n \delta_{i,j} \frac{L_i}{z} \frac{L_j}{z} + \sum_{i=1}^n \sum_{k=1}^{m-1} \gamma_{i,k} \frac{L_i}{z} \ln\left(\frac{w_k}{w_m}\right) + \sum_{t=1}^{T-1} \lambda_t D_t \right] \\ \times \exp\left[ \sum_{k=1}^m \phi_k \ln\left(\frac{w_k}{w_m}\right) + \frac{1}{2} \sum_{k=1}^{m-1} \sum_{l=1}^{m-1} \varphi_{k,l} \ln\left(\frac{w_k}{w_m}\right) \ln\left(\frac{w_l}{w_m}\right) \right] + \varepsilon \qquad \dots (1)$$

where  $\pi$  is profit,  $L_i$  is the *i*<sup>th</sup> loan output, i = 1, 2, ..., n (n = 5 in our case). *z* is the fixed netput (total assets),  $w_k$  is the  $k^{\text{th}}$  input price, k = 1, 2, ..., m (m = 3 in our case). For the input prices, we have:  $w_1$  (price of funds, proxied by the ratio of interest expenses to total deposits),  $w_2$  (price of fixed capital, proxied by the ratio of other operating expenses to fixed assets),  $w_3$  (price of labor, proxied by the ratio

<sup>&</sup>lt;sup>10</sup> We choose composite form instead of translog functional form because translog functional forms suffer from several drawbacks when it is applied to measure scope economies. For example, it does not allow zero values for outputs and fixed netputs, the translog is not well behaved in regions around zero, and scope economy estimates have been shown to vary widely, depending upon the value chosen to represent specialized production (Berger, et al. (1987), Roller (1990)).

of personnel expenses to total number of employees).  $D_t$  is the  $t^{th}$  year dummy, where t = 1, 2, ..., T-1 (T = 11 in our case).

In the profit function specified in equation (1), we normalize the dependent variable of the profit and cost functions by the quantity of the fixed netput (*z*) and the price of last input (*w*<sub>3</sub>). We also normalize all the output terms by *z*, and we normalize the first two input prices by *w*<sub>3</sub>. As mentioned earlier, *z* is total assets, and *w*<sub>3</sub> is the price of labor. The normalization by total assets, *z*, is designed to control for heteroskedasticity, reduce scale biases in estimation, and give the model more economic interpretation. The normalization by *w*<sub>3</sub> imposes linear homogeneity in the input prices (see Berger et al. (2000b) for more discussion). In the following, let  $\pi(L_1, L_2, L_3, L_4, L_5, w, z, year)$  and C( $L_1, L_2, L_3, L_4, L_5, w, z, year$ ) be the predicted profit function and cost functions for all variables at their means.<sup>11</sup>

After using nonlinear least squares regressions to estimate the coefficients in the composite profit function based on the observed sample of joint producers, we can now obtain the predicted profits for both observed diversified banks and the hypothetical focused banks, by assuming that the focused banks follow the same profit function as the diversified banks. The predicted profits for observed diversified banks and hypothetical focused banks enable us to measure the profit scope economies, which is defined as the proportional increase in predicted profits from 5 focused banks versus producing jointly, as shown in the following:

$$S_{\Pi}^{T}(1,2,3,4,5) = \{ \pi(L_{1}, L_{2}, L_{3}, L_{4}, L_{5}, w, z, year) - [\pi(L_{1}, 0, 0, 0, 0, w, z/5, year) + \pi(0, L_{2}, 0, 0, 0, w, z/5, year) + \pi(0, 0, L_{3}, 0, 0, w, z/5, year) + \pi(0, 0, 0, 0, L_{4}, 0, w, z/5, year) + \pi(0, 0, 0, 0, 0, L_{5}, w, z/5, year)] \} / \pi(L_{1}, L_{2}, L_{3}, L_{4}, L_{5}, w, z, year). ... (2)$$

Note that in equation (2), z/5 is used when calculating the predicted profits for the hypothetical

<sup>&</sup>lt;sup>11</sup> Note that we estimate the alternative profit function, which specifies output quantities, rather than the standard profit function, which specifies output prices.

focused firms because we assume that each of the 5 focused firms only has z/5 assets instead of z assets. As discussed elsewhere (e.g., Berger et al. (2000b)), some problems can arise when estimating scope economies, including inaccurate evaluations of specialist costs and inaccurate measurement of scope economies are introduced because considerable extrapolation beyond the sample data is required given that no firms are observed at zero outputs. Therefore, quasi-scope economies may be calculated, assuming that firms produce at least the minimum observed value of each of the outputs (and we term such hypothetical focused firm as the 'hypothetical quasi-focused firm'):

$$S_{\Pi}^{T}(1,2,3,4,5) = \{ \pi(L_1, L_2, L_3, L_4, L_5, w, z, year) - [\pi(L_1 - 4L_{1\min}, L_{2\min}, L_{3\min}, L_{4\min}, L_{5\min}, w, L_{5\max}, w,$$

z/5, year)

$$+ \pi(L_{1\min}, L_2 - 4L_{2\min}, L_{3\min}, L_{4\min}, L_{5\min}, w, z/5, year)$$

$$+ \pi(L_{1\min}, L_{2\min}, L_3 - 4L_{3\min}, L_{4\min}, L_{5\min}, w, z/5, year)$$

$$+ \pi(L_{1\min}, L_{2\min}, L_{3\min}, L_4 - 4L_{4\min}, L_{5\min}, w, z/5, year)$$

$$+ \pi(L_{1\min}, L_{2\min}, L_{3\min}, L_{4\min}, L_5 - 4L_{5\min}, w, z/5, year)$$

$$+ \pi(L_1, L_2, L_3, L_4, L_5, w, z, year).$$

$$\dots (3)$$

Analogous to profit scope economies measured by equation (2) and/or (3), we present a new measure here and term it as 'economies of diversification,' which is described below in equation (4) – (6). We formally define economies of diversification as the increase in profits or reduction in costs of firms diversifying across various products versus specializing in each of the products. The difference between our new measure here and more conventional measure of economies of scope (described by equation (2) and (3)) lies in the differences of the assumptions: We assume that the hypothetical focused bank pools together all the resources into producing one of the products at one time (so that  $\sum_{i=1}^{n} L_i$  is produced for each of the products at one time for the focused bank, where n is the total number

of products for a diversified bank), while the conventional measures assume the hypothetical focused

bank only produces  $L_i$  in product *i* at one time. Therefore, in our context of the product diversification for the Chinese banks, our new approach yields five measures, each measures the proportional differences in predicted profits between the observed diversified bank and a hypothetical focused bank which has loans of one type, but of the same total value of loans as the diversified bank. For the second loan category, for example, the measure would be:

$$D_{\Pi}^{T}(1,2,3,4,5) = [\pi(L_{1}, L_{2}, L_{3}, L_{4}, L_{5}, w, z, year) - \pi(0, L_{1} + L_{2} + L_{3} + L_{4} + L_{5}, 0, 0, 0, w, z, year)]$$

$$/\pi(L_1, L_2, L_3, L_4, L_5, w, z, year).$$
 ... (4)

We also use this measure for some different portfolios that are observed in the data set. Again, to deal with the problem of zero outputs, we create the measure of quasi-diversification economies, where the focused firms produce at least the minimum of each output. For the second product, for example, the measure would be:

$$QD_{\Pi}^{T}(1,2,3,4,5) = \{\pi(L_{1}, L_{2}, L_{3}, L_{4}, L_{5}, w, z, year) - \pi[L_{1min}, (L_{1} + L_{2} + L_{3} + L_{4} + L_{5}) - (L_{1min} + L_{3min} + L_{4min} + L_{5min}), L_{3min}, L_{4min}, L_{5min}, w, z, year]\}$$
  
/  $\pi(L_{1}, L_{2}, L_{3}, L_{4}, L_{5}, w, z, year).$  ... (5)

We also give the quasi-diversification economies for cost measures as below:

$$QD_{C}^{T}(1,2,3,4,5) = \{ C[L_{1\min}, (L_{1} + L_{2} + L_{3} + L_{4} + L_{5}) - (L_{1\min} + L_{3\min} + L_{4\min} + L_{5\min}), L_{3\min}, L_{4\min}, L_{5\min}, w, z, year] - C(L_{1}, L_{2}, L_{3}, L_{4}, L_{5}, w, z, year) \} / C(L_{1}, L_{2}, L_{3}, L_{4}, L_{5}, w, z, year).$$
(6)

The measurement of profit economies of diversification for deposit, asset, and geographical portfolios is calculated in a similar fashion. In all cases, a positive number suggests a diversification premium or economies of diversification, and a negative number suggests a diversification discount, or diseconomies of diversification.

Some ambiguity can potentially arise regarding the specific parameters that can be directly linked to the observed diversification premium or discount here. In other words, it is the combined effect of number of portfolio categories, share in each category, and the estimated profit (or cost) function that results in the difference of profit premium (or cost discount) that we derive for each of the bank-year observations in our sample.

#### 4.3 Focus Index

We also construct a more conventional measure of focus versus diversification, the Focus Index. We measure the Focus Index by employing a Herfindahl-Hirschman Index (HHI) measure following Acharya et al. (2006). The Focus Index is the sum of squares of the proportions of portfolios in each classification. In our case, we construct four different Focus Indices, which are loan, deposit, asset, and geographic Focus Indices.

In the case of loan Focus Index, if we denote the loans in each of the five loan categories as  $L_i$ , where i =1, 2, ..., 5, we then have:

Loan Focus Index = 
$$\sum_{i=1}^{5} (L_i / Q)^2$$
, where  $Q = \sum_{j=1}^{5} L_j$ . ...(7)

Note that the Focus Index, by definition, ranges from 1/n, or 1/5 in this case, to 1, with higher a value of the index indicating more focus (less diversification). We also acknowledge that the Focus Indices suffer an arbitrary feature that is shared by all HHI approach-based measures in that they are incapable of capturing the difference in dimensions of diversification when same level of Focus Indices can be achieved by either changing the number of categories (e.g., loan types) or changing the proportions (e.g., the shares of loans in each category).

#### *4.4 Efficiency measures*

Cost and profit efficiency measure how well a bank is predicted to perform relative to a "bestpractice" bank producing the same outputs under the same environmental conditions. That is, efficiency measures how close to the minimum cost or maximum profit a bank is, where the minimum and maximum are determined by best performers in the sample. We estimate efficiency levels by specifying the commonly-used translog functional form for the cost and profit functions. For convenience, we show only the cost function:

$$\ln(C/w_{3}z)_{it} = \alpha_{0} + \sum_{j=1}^{4} \delta_{j} \ln(y_{j}/z)_{it} + \frac{1}{2} \sum_{j=1}^{4} \sum_{k=1}^{4} \delta_{jk} \ln(y_{j}/z)_{it} \ln(y_{k}/z)_{it} + \sum_{l=1}^{2} \beta_{l} \ln(w_{l}/w_{3})_{it} + \frac{1}{2} \sum_{l=1}^{2} \sum_{m=1}^{2} \beta_{lm} \ln(w_{l}/w_{3})_{it} \ln(w_{m}/w_{3})_{it} + \sum_{j=1}^{4} \sum_{l=1}^{2} \theta_{jl} \ln(y_{j}/z)_{it} \ln(w_{l}/w_{3})_{it}$$

+ year dummies +  $\ln \mu_{it}$  +  $\ln \nu_{it}$  ...(8)

where *i*, *t* index the bank and year, respectively,  $k = 1, \dots 4$  index the four output variables, and  $\delta_{jk} \equiv \delta_{kj}$ . *C* represents the bank' s total costs. There are four outputs (y): total loans, total deposits, liquid assets, other earning assets; three input prices (w):  $w_I$  (price of funds, proxied by the ratio of interest expenses to total deposits),  $w_2$  (price of fixed capital, proxied by the ratio of other operating expenses to fixed assets),  $w_3$  (price of labor, proxied by the ratio of personnel expenses to total number of employees); and one fixed netput (z): total assets. The  $\ln \mu_u$  term represents a bank's efficiency level and  $\ln v_u$  is a random error that incorporates both measurement error and luck. The cost function is estimated using the  $\ln \mu_u + \ln v_u$  as a composite error term. The normalization by bank's total assets (z) reduces heteroskedasticity, and allows banks of any size to have comparable residual terms from which the efficiencies are calculated. The normalization by the last input price ( $w_3$ ) ensures price homogeneity.

A bank's cost efficiency score is determined by comparing its actual costs to best-practice

minimum costs to produce the same output under the same conditions using estimates of the efficiency factor  $\ln \mu_{it}$ , which is disentangled from the estimated cost function residual using half-normal distributional assumptions, whereas  $\ln \nu_{it}$  follows the normal distribution. Profit efficiency scores are estimated similarly. Total profits replace total costs and we add a constant before taking the log to avoid taking a log of negative number.

#### 4.5 *Other variables*

We employ the following (annual) variables obtained from Bankscope and other sources for the banks in our sample over the period 1996-2006:

#### Performance measures:

- ROA: return on assets measured as the ratio of net income to total assets.<sup>12</sup>
- Expenses/assets: the ratio of total expenses to total assets.

#### Risk measures:

- Loan loss provisions/assets: the ratio of loan loss provisions to total assets, which can also be interpreted as an ex ante measure of the level of expected losses.<sup>13</sup>
- Nonperforming loans/assets: the ratio of nonperforming loans to total assets, which can be interpreted as an ex post measure of the actual losses from lending activities.
- Equity/assets: capital ratio of the bank measured as equity (book value)/assets, the approximate equivalent of the bank's tier 1 capital ratio.

#### Size measures:

We use a continuous variable, i.e., ln(assets) as the measure for size of the bank. The

<sup>&</sup>lt;sup>12</sup> For the foreign banks in our sample, we use the ROA of the Chinese operations, not the consolidated earnings of these banks.

<sup>&</sup>lt;sup>13</sup> We also tried to use alternative proxy for risk, i.e., ROA volatility, which is measured as the standard deviation of ROA over the sample years for the same bank, and the results remain qualitatively the same.

continuous variable such as ln(assets) is normally expected to be a superior regressor than some arbitrary size dummies, except the case when there is a non-monotonic relationship between size and performance. Therefore, in all regression analyses of the paper, we also include the squared term of ln(assets) to control for the potential nonlinear relationship between size and performance.<sup>14</sup>

#### Ownership variables:

In this paper, all ownership variables are measured at the bank-year level. First, we define a dummy variable, Big Four, to represent the four biggest Chinese banks, i.e., Bank of China, Industrial and Commercial Bank of China, Agricultural Bank of China, and China Construction Bank. Second, we construct several continuous variables to capture more nuances of the ownership structure of the Chinese banks. State ownership represents the percentage of shares owned by the state and state-owned enterprises (other than the Big Four). Domestic private ownership measures the percentage of shares owned by domestic private individuals and institutions. Foreign ownership captures the percentage of shares owned by foreigners (foreign individuals and institutions).<sup>15</sup>

#### *Conglomerate affiliation:*

Conglomerate affiliation dummy equals 1 if the bank is affiliated with a conglomerate, 0 otherwise.

#### *Hyper-regulated period:*

Hyper-regulated period dummy equals 1 during 1996-2001 period, 0 otherwise.

#### 4.6 Regression analysis

There are two sets of regression analyses in our study. First, we conduct a regression analysis

<sup>&</sup>lt;sup>14</sup> We thank an anonymous referee for this suggestion. We provide discussion of some further robustness tests using size dummies at the end of Section 5.

<sup>&</sup>lt;sup>15</sup> We thank an anonymous referee for suggesting that we use the continuous ownership variables instead of discretized ownership data in the analysis. In addition, we provide some discussion of additional robustness tests in which we use those ownership dummy variables in the regressions at the end of Section 5.

of the determinants of the economies of diversification, where the dependent variables are profit premiums and cost discounts. The profit premium is the measure shown in equation (5) above – the proportional difference in predicted profits between the observed diversified bank and a hypothetical quasi-focused firm that produces the same total amount of output, while the hypothetical quasi-focused firm is defined earlier in Section 4.2 as the hypothetical focused firm which produces at least the minimum of each of the products. The cost discount (shown in equation (6)) is analogously defined as the proportional difference in predicted costs of a quasi-focused firm and the observed diversified bank. The profit premium and cost discount are regressed on a group of bank-level characteristics including ownership variables, risk measures, and bank size. We are particularly interested in the effects of foreign ownership, which was shown to be an important determinant of performance of Chinese banks in an earlier study (Berger et al., 2009). Risk and bank size are included in the regressions as control variables. In the second regression analysis, we regress accounting performance measures (i.e., ROA and costs/assets), and stochastic frontier efficiency measures on the more conventional measures of diversification – the Focus Indices that we construct (including loan, deposit, asset and geographic Focus Indices), along with the ownership, risk, and size.

#### 5. Empirical Results

Table 1 presents summary statistics for ROA, the cost ratio, decomposition of loan, deposit, asset, and geographical portfolios, and also some firm-level characteristics variables, including risk proxies and size variables – these variables follow the definitions provided in Section 4. All financial items are in billions of \$US, and inflation-adjusted to the price level of year 1996. This table shows that, among the components of loan portfolios of an average Chinese bank, industry loans make up the biggest proportion of the total loans (67.10%), followed by commercial loans (14.88%), consumer loans (7.74%), real estate loans (5.87%), and agriculture loans (4.41%), while the shares of each

components change significantly from one bank to another. Among the components of deposit portfolios of an average Chinese bank, customer demand deposits make up the largest share (56.76%), followed by customer savings deposits (34.36%), banking deposits (8.29%), corporate deposits (0.3%), and other deposits (0.28%). Among the components of asset portfolios, loans make up the largest share (59.45%), followed by financial investments (22.95%), other assets (10.79%), deposits in other banks (5.21%), and total fixed assets (1.60%). In terms of geographical diversification, Eastern China makes up the largest proportion of an average Chinese bank (42.31%), followed by Northern China (24.44%), Southern China (14.16%), Central China (10.76%), and Western China (8.34%).

Table 2 presents summary statistics of the profit premiums and cost discounts measured by economies of diversification for loan portfolios. The definitions of the decompositions of the portfolios are the same as in Table 1. A profit premium is defined as the proportional difference in the predicted profits between the observed diversified bank and a hypothetical quasi-focused bank. A cost discount is analogously defined as the proportional difference in the predicted costs between a hypothetical quasi-focused bank and the observed diversified bank. Table 2 shows the summary statistics for each loan category as well as the average profit premium and cost discount over all of the loan categories.

The negative means of profit premiums and cost discounts suggest that more focus is associated with higher profits and lower costs. The overall mean of the profit premium, -0.203, implies that the diversified banks, on average, are losing about 1/5 of their profits compared to the hypothetical bank which puts almost all its value of loans into one category. The mean of cost discount, -0.117, implies that the diversified banks, on average, could have saved about 1/10 of their costs if they focused on one of the loan categories. The consistency of the negative findings is convincing evidence in favor of the hypothesis that lending focus is associated with better profit and cost performance.

Table 3 presents the summary statistics including the mean, median, standard deviation,

minimum, and maximum for the measures of economies of diversification in terms of profits and costs for all the banks over the sample period 1996-2006. Summary statistics of subsample banks by size class (i.e., small banks, medium banks, large banks) are also presented in Table 3. Panel A of Table 3 presents the summary statistics of the measure of economies of diversification in terms of profit premiums and cost discounts for all observations and all of the individual loan, deposit, asset, and geographic categories. Note that the loan diversification summary statistics match the average loan profit premium and cost discount shown in Table 2. Panels B, C, and D of Table 3 presents the summary statistics of the economies of diversification for small banks, medium banks, and large banks, respectively. All of the means of the profit premiums and cost discounts are negative, and this holds for every size class, which strongly suggests diseconomies of diversification for Chinese banks in terms of their product and geography portfolios. Again, the consistency across product and geographic categories and across size classes strongly suggests that focused firms are more profitable and less costly for a given total output than diversified firms.

Table 4 presents the ordinary least squares (OLS) regressions of the profit premiums on ownership variables and other firm-level characteristics, including risk proxies, size measures, conglomerate affiliation dummy, and hyper-regulated period dummy. As noted, the conglomerate affiliation dummy equals 1 if the bank is affiliated with a conglomerate, 0 otherwise, and the hyper-regulated period dummy equals 1 for period 1996-2001, and 0 otherwise. Domestic private ownership and the less-regulated period are considered omitted variables in the regressions. In this table as well as throughout the analysis of this study, we use the one-year lagged term (i.e., year t-1) of ownership variables instead of concurrent terms of ownership because normally it takes some time for the ownership to affect performance. The profit premiums in the first, second, third, and fourth paired columns are calculated based on decompositions of loan, deposit, asset, and geographical portfolios, respectively, and these decompositions follow the same definitions as described in previous tables. We

present two columns for each definition of portfolio decomposition with two different risk measures, namely, loan loss provisions/assets, and nonperforming loans/assets, entered into the regression alternatively. Loan loss provision/assets captures the ex ante estimation of the expected losses from lending activities, while nonperforming loans/assets is an ex post measure. Table 5 presents the regressions of cost discounts on the same set of variables. While there are a number of results in Table 4 and Table 5, we concentrate our attention on the foreign ownership for two reasons. First, in prior research, foreign bank ownership in China is associated with significantly better profit and cost efficiency (Berger et al., 2009). Second, as will be shown, the results for foreign ownership are consistent across a number of regressions in this paper. Except for the case of geography diversification in the profit premium regression, foreign ownership is associated with significant and positive coefficients. This suggests that foreign owners are able to at least partially mitigate the diseconomies of diversification, making banks suffer fewer profit losses and fewer cost increases associated with diversification.

Table 6 presents the summary statistics of Focus Indices which are based on Herfindahl-Hirschman Index (HHI) measures of loan, deposit, asset, and geographic portfolios. The method by which the Focus Indices are calculated is described in the previous section. By definition, the Focus Indices range from the value of 1/5 to 1, the closer the value to 1 (i.e., the higher the value), the more focused (i.e., less diversified) the bank is. In Table 6, we note that most of the firms are fairly well diversified in terms of loans, deposits, and assets, but are more focused in terms of their geographic region, particularly the small and medium sized institutions.

Table 7 presents the ordinary least squares (OLS) regressions of return on assets (ROA) on the Focus Indices, ownership variables, conglomerate affiliation dummy, the interaction terms between the Focus Indices and ownership, between Focus Indices and conglomerate affiliation dummy, and control variables including risk proxies, size measures, and hyper-regulated period dummy. Domestic private ownership and the less-regulated period are considered omitted variables in the regressions. The Focus Index in the first pair of columns represents the loan Focus Index, and the same Focus Index is also used to construct the interaction terms between ownership (and conglomerate affiliation) and Focus Indices. The Focus Indices in the second, third, and fourth pairs of columns represent deposit, asset, and geographic Focus Indices, respectively. <sup>16</sup> To partially address the issue concerning the endogeneity of focus measures, we consider the relationship between focus in year t-1 on performance measures in year t. We complement this analysis with a robustness check that employs focus measures in year t as well, but treats them as endogenously determined variables. These tests show that our results are robust to the endogeneity of focus measures.<sup>17</sup>

We call attention to three main results in Table 7. First, the Focus Indices, across alternative definitions, are positively and significantly associated with ROA, implying that focused banks associated with domestic private ownership enjoy higher levels of ROA on average.<sup>18</sup> When we combine this evidence with the observation of insignificant coefficients of interaction terms between Focus Indices and Big Four, non-Big Four state ownership in this table, it indicates that focused banks – unless they are associated with foreign ownership – yield higher ROA on average. This result is consistent with the finding using the economies of diversification measure in Table 4 and 5, and reinforces the observation there that more focused banks tend to be more profitable than diversified institutions. Second, foreign ownership is positively and significantly associated with ROA, implying that banks with foreign ownershipenjoy higher ROA on average. This is consistent with the higher

<sup>&</sup>lt;sup>16</sup> By treating potential interdependencies of various diversification or focus in terms of their impact on performance separately, we are likely to ignore the "spatial differences," in the fact that there may be closer relationships between some regional pairs, but not others. Similarly, some banks may get deposits and loans, but do not offer other non-traditional facilities. We acknowledge such shortcomings in the tests of the paper, and we thank an anonymous referee for pointing this out.

<sup>&</sup>lt;sup>17</sup> We thank the anonymous referees for this suggestion. The robustness test tables are available upon request.

<sup>&</sup>lt;sup>18</sup> We also run some robustness tests to examine whether the same results hold when we enter the different dimensions of diversification measures (i.e., the different Focus Indices) in the same regression. The results show that our findings on the individual diversification measure are robust after we control for other measures of diversification. These tables are available upon request.

profit efficiency for foreign-owned banks in China found in Berger et al. (2009). Third, the interaction terms between the foreign ownership and Focus Indices are negative and significant, implying that foreign ownership may play a mitigating role when banks diversify, in the sense that the diversified banks are not penalized as much in terms of ROA if they are associated with more foreign ownership. In the same fashion, the negative and significant coefficients of interaction terms between Focus Indices and conglomerate affiliation dummy indicate that conglomerate affiliation also mitigates the negative effects of diversification on ROA.

Table 8 presents the ordinary least squares (OLS) regressions of total expenses/total assets on the same group of independent variables as in Table 7. When we switch our attention to the cost perspective as this table presents, we generally find consistent results. First, the negative and significant coefficients of Focus Indices imply that focused banks enjoy lower costs than diversified banks, holding other factors constant. Banks associated with more foreign ownership, on average, enjoy lower costs than otherwise. The interaction terms between foreign ownership (and conglomerate affiliation) and Focus Indices are generally associated with positive and significant coefficients, which once again imply the mitigating role of foreign ownership (conglomerate affiliation), in that the diversified banks (i.e., banks with lower Focus Indices) are associated with lower costs when foreign ownership (conglomerate affiliation) is in place.

To obtain a more comprehensive understanding of the effect of a bank's focus/diversification strategy, we need to study its effect on both bank profit/cost performance and bank risk.<sup>19</sup> If focus produces an increase in bank profit/cost performance and a decrease in bank risk, then we can interpret this result as implying that focus improves overall bank performance. However, if focus produces an increase in profit/cost performance and an increase in risk, then the overall effects of focus/diversification on banks are ambiguous and cannot be determined without taking a stand on what

<sup>&</sup>lt;sup>19</sup> We thank an anonymous referee for this suggestion.

constitutes an "efficient" risk-return trade-off. Table 9 examines the effects of diversification on the risk of the Chinese banks and it presents the ordinary least squares (OLS) regressions of loan loss provisions/assets (on the left four columns) and nonperforming loans/assets (on the right four columns) on Focus Indices, ownership variables, conglomerate affiliation dummy, the interaction terms between Focus Indices and ownership, between Focus Indices and conglomerate affiliation dummy, and control variables including equity ratio, size measures, and hyper-regulated period dummy. Domestic private ownership and the less-regulated period are omitted variables in the regressions. The Focus Index in the first column represents the loan Focus Index, and the same Focus Index is used to construct the interaction terms between ownership (and conglomerate affiliation) and Focus Indices. The Focus Indices in the second, third, and fourth columns represent deposit, asset, and geographic Focus Indices, respectively. As discussed earlier, loan loss provisions ratio is an ex ante measure of risk while nonperforming loans ratio is an ex post measure of risk. Despite such differences in these two measures, both tables show qualitatively similar results: focused banks, on average, are associated with lower risks, and foreign ownership and conglomerate affiliation tend to mitigate the negative role of diversification on bank risks.

In addition, we apply heteroskedasticity tests to our data in order to investigate in more detail of the issue of the effects of diversification on risk. Recently, some studies (see, e.g., Adams et al. (2005), Cheng (2008)) have applied the Glejser's (1969) heteroskedasticity test in their studies of the variability of performance or risks. To conduct the Glejser test, we first need to specify a model for the performance measure, i.e., ROA. In other words, the first-stage regression includes a number of the right-hand-side variables which could explain performance levels, and in our case, we use the same specification as in Table 7 as in our first-stage regressions. In the second-stage regressions, the test is applied to the absolute values of the residuals of the first regression, which are now treated as the dependent variable, and the independent variables are the variables that could potentially explain the performance volatility. In our test, we include the ownership variables, Focus Indices, interactions between ownership and Focus Indices, equity ratio, size measures, conglomerate affiliation dummy, and hyper-regulated period dummy. The results show that diversification is not significantly associated with an increase in performance volatility, which provides consistent evidence as presented by Table 9. A table of these tests is available upon request. <sup>20</sup>

Table 10 presents the summary statistics of basic variables used in the profit and cost efficiency estimations. In the translog-based estimations of profit (cost) efficiency, output variables considered are total loans, total deposits, liquid assets, and other earning assets, and the input variables are:  $w_1$  (price of funds, proxied by the ratio of interest expenses to total deposits),  $w_2$  (price of fixed capital, proxied by the ratio of other operating expenses to fixed assets),  $w_3$  (price of labor, proxied by the ratio of personnel expenses to total number of employees). The outputs are normalized by total assets, which are considered as a fixed netput z. As above, all financial values are inflation-adjusted to the base year 1996.

Table 11 presents the ordinary least squares (OLS) regressions of profit efficiency on the same group of independent variables as in Table 7. The three main findings using the accounting measures, i.e., ROA and costs/assets, are essentially replicated here. The coefficients of Focus Indices across alternative definitions of diversification are positive and statistically significant, implying that focused banks are associated with greater profit efficiency. Also, the coefficients of foreign ownership are positive and significant, suggesting that foreign ownership is associated with higher profit efficiency. The negative coefficients of the interaction terms between the foreign ownership (conglomerate affiliation dummy) and Focus Indices further confirm the mitigating effects of foreign ownership (conglomerate affiliation) as found in the previous tables.

Table 12 presents the ordinary least squares (OLS) regressions of cost efficiency on the same

<sup>&</sup>lt;sup>20</sup> We thank an anonymous referee for the suggestion of these tests.

group of independent variables as in Table 11. The results again suggest that 1) more focus is associated with higher efficiency; 2) foreign ownership is generally positively related to efficiency; and 3) foreign ownership plays a role in mitigating the poor performance effects of diversification.

We also conduct a number of robustness tests. First, we replace the continuous ownership variables (i.e., state ownership, domestic private ownership, foreign ownership) with the ownership dummies, for the reason that treating ownership as dummies instead of continuous variables has been quite a standard procedure in the banking literature (see, e.g., Saunders (1990), Altunbas (2005), Berger et al. (2005, 2008), Micco et al. (2007)). We define majority state-owned banks as those where the state and state-owned enterprises ownership is > 50% of total ownership (other than the Big Four), majority private domestic banks as those whose private domestic ownership is > 50% of total ownership, majority foreign banks as those whose foreign ownership is > 50% of total ownership, and minority foreign banks as those with 0 < foreign ownership  $\leq 50\%$ . Our robustness test tables show that both majority and minority foreign ownership dummies are associated with the better profit and/or economies of diversification, higher ROA and/or lower costs, lower risk, and higher profit and cost efficiency measures. More importantly, we find both majority and minority foreign ownership dummies play a mitigating role when banks diversify, and these results are essentially consistent with our evidence found with the continuous ownership variables. Second, we replace the continuous size variables with the size dummies (bank size dummies are based on total assets (inflation-adjusted to 1996): the bank is small if assets  $\leq$  US \$1 billion; medium if US \$1 billion < assets  $\leq$  US \$20 billion; and large if assets > US \$20 billion), and if the size is non-monotonically related with performance or risk, the size dummies are expected to capture some extent of such nonlinearity as well. For example, in Table 4, 5, 8, 11, 12, we find that the squared terms of ln(assets), at least for some of the regressions, are associated with significant coefficients. Such convexity (positive coefficients) or concavity (negative coefficients) of size effect on performance found in these tables based on continuous size

variables are not always perfectly captured by the size dummies in our robustness tests, which may be resulted from the fact that the size dummies are constructed based on some rather arbitrary benchmarks rather than the real threshold or discontinuity in the relationship between size and performance of banks. To conclude, our robustness tests show that our main evidence in the paper is robust to different specification of variables and/or assumptions. These tables are available upon request.<sup>21</sup>

#### 6. Discussion and Conclusions

The paper investigates the effects of product and geographical focus and diversification strategies on performance using a sample of 88 Chinese banks representing about 90% of commercial banking assets during 1996-2006. We present a new measure of economies of diversification and we compare the performance effects with those of a more conventional focus index measure. Consistent across these two approaches, and across accounting performance measures and efficiency frontier measures, we find diseconomies of diversification among the sample banks in the loan, deposit, asset, and geographic dimensions. That is, more focused banks are associated with higher profits, lower costs, higher profit efficiency, and higher cost efficiency. The diversification discount evidence that we find for the average Chinese bank in our sample, and their cross-sectional changes across different ownership types as well as their time-series changes across different regulatory regimes, may shed light on the mechanisms as to why Chinese banks are found to suffer the diversification discount and why they still diversify despite the discount. We believe the diversification discount of Chinese banks, at least partially, comes from the lack of managerial expertise of the top management teams and the slack or ineffective incentive schemes for the managers to maximize the shareholders' wealth. In China, most of the banks' managers are appointed by the government, and their promotions largely depend on how well they cooperate with the government's directions instead of making independent

<sup>&</sup>lt;sup>21</sup> We thank the anonymous referees for the suggestion of these tests.

decisions. Under such circumstances, managers are less likely to take a firm stand on the diversification/focus strategy that maximizes the profit of the bank or wealth of the shareholders. The intensity of influence from the central and local governments is expected to follow a downward trend in the forthcoming years, though the influence might still be quite substantial for those majority state-owned banks.

We also find that foreign ownership plays a mitigating role in diseconomies of diversification. The banks that are associated with more foreign ownership tend to suffer a smaller diversification discount or a lesser loss of profits or increase in costs from diversification. In the context of Chinese banks, the foreign ownership represents a presence of monitoring and delivering of managerial expertise at the top management level. At the same time, the foreign ownership is also often associated with more/better networks, partnerships, and even direct or indirect affiliation with international conglomerates. This evidence is consistent with existing studies that find that the involvement of foreign banks provides an additional certification for the domestic banks, and sometimes the mere presence of foreign bank owners improves the culture and efficiency of the overall banking sector (Hasan and Marton, 2003; Berger et. al., 2009). Our observation that foreign bank ownership and conglomerate affiliation tend to mitigate the diseconomies of diversification in the Chinese banking may be of help in the discussion of policy recommendations with regards to foreign bank entry in China and other emerging markets.

#### 7. References

- Adams, R. B., Almeida, H., Ferreira, D. 2005. Powerful CEOs and their impact on corporate performance. Review of Financial Studies 18, 1403-1432.
- Acharya, V.V., Hasan, I., Saunders, A., 2006. Should banks be diversified? Evidence from individual bank loan portfolios. Journal of Business 79, 1355-1412.
- Altunbas, Y., Evans L., and Molyneux, P., 2001. Bank ownership and efficiency, Journal of Money, Credit and Banking 33, 926-954.
- Amihud, Y., Lev, B., 1981. Risk reduction as a managerial motive for conglomerate mergers, Bell Journal of Economics 12, 605–617.
- Berger, A.N., Clarke G., Cull, R., Klapper, L.F., Udell, G.F., 2005. Corporate governance and bank performance: A joint analysis of the static, selection, and dynamic effects of domestic, foreign, and state ownership. Journal of Banking & Finance 29, 2179-2221.
- Berger, A.N., Cummins, J.D., Weiss, M.A., Zi, H., 2000b. Conglomeration versus strategic focus: Evidence from the insurance industry. Journal of Financial Intermediation 9, 323-362.
- Berger, A.N., Demsetz, R., Strahan, P., 1999. The consolidation of financial services industry: Causes, consequences, and implications for the future. Journal of Banking and Finance 23, 135-194.
- Berger, A.N., DeYoung, R., Genay, H., Udell, G.F., 2000a. The globalization of financial institutions: Evidence from cross-border banking performance. Brookings-Wharton Papers on Financial Services 3, 23-158.
- Berger, A.N., Hancock, D., Humphrey, D.B., 1993. Bank efficiency derived from the profit function. Journal of Banking and Finance 17, 317-347.
- Berger, A.N., Hanweck, G.A., Humphrey, D.B., 1987. Competitive viability in banking: Scale, scope, and product mix economies. Journal of Monetary Economics 20, 501-520.

- Berger, A.N., Hasan, I., Zhou, M., 2009. Bank ownership and efficiency in China: What will happen in the world's largest nation? Journal of Banking and Finance 33, 113-130.
- Berger, A.N., Humphrey, D.B., 1991. The dominance of inefficiencies over scale and product mix economies in banking. Journal of Monetary Economics 28, 117-148.
- Berger, A.N., Humphrey, D.B., Pulley, L.B., 1996. Do consumers pay for one-stop banking? Evidence from an alternative revenue function. Journal of Banking and Finance 20, 1601-1621.
- Berger, A.N., Klapper, L.F., Martinez Peria, M.S., Zaidi, R., 2008. Bank ownership type and banking relationships. Journal of Financial Intermediation 17, 37-62.
- Berger, P.G., Ofek, E., 1995. Diversification's effect on firm value. Journal of Financial Economics 37, 39-65.
- Berger, P.G., Ofek, E., 1996. Bustup takeovers of value-destroying diversified firms. The Journal of Finance 51 (4), 1175-1200.
- Boot, A., Schmeits, A., 2000. Market discipline and incentive problems in conglomerate firms with applications to banking, Journal of Financial Intermediation 9, 240–273.
- Boyd, J.H., Graham, S.L., 1988. The profitability and risk effects of allowing bank holding companies to merge with other financial firms: A simulation study, Federal Reserve Bank of Minneapolis Quarterly Review, 476-514.
- Boyd, J.H., Prescott, E., 1986. Financial intermediary coalitions. Journal of Economic Theory 38, 211– 32.
- Cerasi, V., Daltung, S., 2000. The optimal size of a bank: Costs and benefits of diversification. European Economic Review 44, 1701-1726.
- Cheng, S. 2008. Board size and the variability of corporate performance. Journal of Financial Economics 87, 157-176.

- Clark, J., Siems, T.F., 1997. Competitive viability in banking: Looking beyond the balance sheet. Federal Reserve Bank of Dallas, Financial Industry Studies Working Paper No. 5-97.
- DeLong, G., 2001. Stockholder gains from focusing versus diversifying bank mergers. Journal of Financial Economics 59, no 2:221-52.
- Demsetz, R.S., Strahan, P.E., 1997. Diversification, size and risk at bank holding companies. Journal of Money, Credit and Banking 29, 300-313.
- Deng, S., Elyasiani, E., 2007. Geographic diversification and BHC return and risk performance. Working paper.
- Denis, D.J., Denis, D.K., Sarin, A., 1997. Agency problems, equity ownership, and corporate diversification. Journal of Finance 52, 135-160.
- DeYoung, R., Roland, K.P., 2001. Product mix and earnings volatility at commercial banks: Evidence from a degree of leverage model. Journal of Financial Intermediation 10 (1), 54-84.
- Diamond, D., 1984. Financial intermediation and delegated monitoring. Review of Economic Studies 59, 393-414.
- Drucker, S., Puri, M., 2009. On loan sales, loan contracting, and lending relationships, Review of Financial Studies, forthcoming.
- Fama, E., 1980. Banking in the theory of finance. Journal of Monetary Economics 9, 39–58.
- Fama, E., 1985. What's different about banks? Journal of Monetary Economics 15, 29–39.
- Fauver, L., Houston, J.F., Naranjo, A., 2004. Cross-country evidence on the value of corporate industrial and international diversification, Journal of Corporate Finance 10, 729-752.
- Glejser, H. 1969. A new test for heteroskedasticity. Journal of the American Statistical Association 64, 316-323.
- Goldberg, L.G., Hanweck, G.A., Keenan, M., Young, A., 1991. Economies of scale and scope in the securities industry. Journal of Banking and Finance 15, 91-107.

- Hasan, I., and Marton, K. 2003. Development and efficiency of the banking sector in a transition economy: Hungarian evidence. Journal of Banking and Finance, 27, 2249-2270.
- Hunter, W. C., Timme, S. G. and Yang, W. K., 1990. An examination of cost subadditivity and multiproduct production in large US banks. Journal of Money, Credit and Banking 22, 504-525.
- Iskandar-Datta, M., McLaughlin, R., 2005. Global diversification: New evidence from corporate operating performance, Working Paper, retrieved from www.fma.org/chicago/papers/fmadiversity.pdf.
- James, C., 1987. Some evidence on the uniqueness of bank loans. Journal of Financial Economics 19, 217–36.
- Jensen, M., 1986. Agency costs of fee cash flow, corporate finance, and takeovers. American Economic Review 76, 323-329.
- Khanna, T., Yafeh, Y., 2005, Business Groups and Risk Sharing around the World, The Journal of Business 78, 301-340.
- Khanna, T., Yafeh, Y., 2007, Business groups in emerging markets: Paragons or parasites? Journal of Economic Literature 45, 331-372.
- Klein, P., Saidenberg, M., 1998. Diversification, organization, and efficiency: Evidence from bank holding companies. Working paper. Federal Reserve Bank of New York.
- Laeven, L., Levine, R., 2007. Is there a diversification discount in financial conglomerates? Journal of Financial Economics 85, 331-367.
- Lamont, O.A., Polk, C., 2002. Does diversification destroy value?: Evidence from industry shocks, The Journal of Financial Economics 63, 51-77.
- Lang, L.H.P., Stulz, R.M., 1994. Tobin's Q, corporate diversification, and firm performance, Journal of Political Economy 102, 1248-1280.

- Lown, C.S., Osier, C.L., Strahan, P.E., Sufi, A., 2000. The changing landscape of the financial service industry: What lies ahead? Economic Policy Review, Federal Reserve Bank of New York 6 (October 2000), 39-54.
- Mahajan, A., Rangan, N., Zardkoohi, A., 1996. Cost structures in multinational and domestic banking. Journal of Banking and Finance 20, 238-306.
- McKillop, D. G., Glass, J. C., Morikawa, Y., 1996. The composite cost function and efficiency in giant Japanese banks, Journal of Banking and Finance 20, 1651–1671.
- Mester, L.J., 1987. A multiproduct cost study of savings and loans. Journal of Finance 42, 423-445.
- Mester, L. J., 1993. Efficiency in the savings and loan industry. Journal of Banking and Finance 17, 267-86.
- Micco, A., Panizza, U. Yañez, M., 2007. Bank ownership and performance. Does politics matter? Journal of Banking and Finance 31, 219-241.
- Miller, S., Parkhe, A., 2002. Is there a liability of foreignness in global banking? An empirical test of banks' X-efficiency. Strategic Management Journal 23, 55-75.
- Noulas, A.G., Miller, S.M., Ray, S.C., 1990. Returns to scale and input substitution for large U.S. banks. Journal of Money, Credit, and Banking 22, 94-108.
- Odesanmi, S., Wolfe, S., 2007. Can revenue diversification reduce systemic risk? Empirical evidence from banks in emerging economies, University of Southampton, School of Management Working paper.
- Pulley, L.B., Braunstein, Y.M., 1992. A composite cost function for multiproduct firms with an application to economies of scope in banking. The Review of Economics and Statistics 74, 221-230.

- Pulley, L. B., Humphrey, D.B., 1993. The role of fixed costs and cost complementarities in determining scope economies and the cost of narrow banking proposals. Journal of Business 66, 437-462.
- Rajan, R., 1992. Insiders and outsiders: The choice between relationship and arms length debt. Journal of Finance 47, 1367–1400.
- Ramakrishnan, R., Thakor, A., 1984. Information reliability and a theory of financial intermediation. Review of Economic Studies 51, 415-432.
- Robinson, K., 2007. For mainland Chinese banks, overseas acquisitions require more than money. The International Herald Tribune, August 6, 2007.
- Roller, L. 1990. Proper quadratic cost functions with an application to the Bell system. Review of Economics and Statistics 72, 202-210.
- Rose, P.S., 1989. Diversification of the banking firm. Financial Review 24, 251-280.
- Saunders, A., Strock, E., Travlos, N.G., 1990. Ownership structure, deregulation, and bank risk taking, The Journal of Finance 45, 643-654.
- Servaes, H., 1996. The value of diversification during the conglomerate merger wave. Journal of Finance 51, 1201-1225.
- Sharpe, S., 1990. Asymmetric information, bank lending, and implicit contracts: A stylized model of customer relationships. Journal of Finance 55, 1069–87.
- Stiroh, K.J., 2004. Diversification in banking: Is noninterest income the answer? Journal of Money, Credit and Banking 36, 853-82.
- Templeton, W.K., Severiens, J.T., 1992. The effects of non-bank diversification on bank holding company risk. Quarterly Journal of Business and Economics 31, 3-17.
- Villalonga, B., 2004, Diversification discount or premium? New evidence from the Business Information Tracking Series, The Journal of Finance 59, 479-506.

- Winton, A., 1999. Don't put all your eggs in one basket? Diversification and specialization in lending.Working paper, University of Minnesota.
- Yuan, W., 2006, Corporate banking in China: History, opportunities and challenges, Report Published by Celent.

#### Table 1 Summary statistics of firm-level characteristics variables

Table 1 presents summary statistics of ROA, cost ratio, decomposition of loan, deposit, asset, and geographical portfolios, and also some firm-level characteristics variables, including risk proxies and size variables. The disaggregated loan decomposition is based on each bank's loans to industry loans, commercial loans, real estate loans, agriculture loans, and consumer loans. The disaggregated deposit decomposition is based on each bank's customer demand deposits, customer savings deposits, non-bank corporate deposits, deposits by other banks, and other deposits. The disaggregated asset decomposition is based on each bank's total loans, deposits in other banks, financial investments (including investment in government securities, trading securities, and other financial assets), total fixed assets, and other assets. The disaggregated geographical decomposition is based on each bank's loans on various regions in China, including northern, central, eastern, southern, and western regions. Conglomerate affiliation equals to 1 if the bank is affiliated with a conglomerate, 0 otherwise. Hyper-regulated period equals to 1 during 1996-2001 period, 0 otherwise. All the financial items, except ln(assets, in thousand \$US), are in billions of \$US, and inflation-adjusted to the price level of year 1996.

| Variable                    | Obs. | Mean   | Median | Std Dev. | Minimum | Maximum  |
|-----------------------------|------|--------|--------|----------|---------|----------|
| Performance measures        |      |        |        |          |         |          |
| Return on assets (ROA)      | 464  | 0.016  | 0.014  | 0.006    | -0.003  | 0.100    |
| Total expenses/assets       | 464  | 0.025  | 0.020  | 0.014    | 0.004   | 0.087    |
| Loan decomposition          |      |        |        |          |         |          |
| Industry loans              | 464  | 18.000 | 0.979  | 54.900   | 0.002   | 416.000  |
| Commercial loans            | 464  | 3.991  | 0.240  | 13.900   | 0.000   | 141.000  |
| Real estate loans           | 464  | 1.576  | 0.029  | 6.897    | 0.000   | 78.200   |
| Agriculture loans           | 464  | 1.183  | 0.084  | 5.828    | 0.000   | 89.500   |
| Consumer loans              | 464  | 2.076  | 0.262  | 6.608    | 0.000   | 70.700   |
| Deposit decomposition       |      |        |        |          |         |          |
| Customer demand deposits    | 464  | 22.300 | 1.615  | 67.200   | 0.000   | 423.000  |
| Customer savings deposits   | 464  | 13.500 | 0.665  | 42.600   | 0.000   | 384.000  |
| Corporate deposits          | 464  | 0.117  | 0.004  | 0.400    | 0.000   | 3.826    |
| Banking deposits            | 464  | 3.258  | 0.135  | 9.717    | 0.000   | 81.100   |
| Other deposits              | 464  | 0.110  | 0.002  | 0.396    | 0.000   | 3.849    |
| Asset decomposition         |      |        |        |          |         |          |
| Loans                       | 464  | 27.200 | 1.684  | 82.100   | 0.000   | 725.000  |
| Deposits in other banks     | 464  | 2.383  | 0.173  | 7.930    | 0.000   | 96.700   |
| Financial investments       | 464  | 10.500 | 0.681  | 35.400   | 0.000   | 341.000  |
| Total fixed assets          | 464  | 0.733  | 0.051  | 2.181    | 0.000   | 12.900   |
| Other assets                | 464  | 4.939  | 0.316  | 14.600   | 0.000   | 98.900   |
| Geographical decomposition  |      |        |        |          |         |          |
| (loans)                     |      |        |        |          |         |          |
| Northern                    | 464  | 6.180  | 0.000  | 89.200   | 0.000   | 1760.000 |
| Central                     | 464  | 2.720  | 0.000  | 37.700   | 0.000   | 648.000  |
| Eastern                     | 464  | 10.700 | 0.000  | 151.000  | 0.000   | 2410.000 |
| Southern                    | 464  | 3.580  | 0.000  | 47.700   | 0.000   | 861.000  |
| Western                     | 464  | 2.110  | 0.000  | 26.800   | 0.000   | 447.000  |
| Control variables           |      |        |        |          |         |          |
| Loan loss provisions/assets | 464  | 0.013  | 0.010  | 0.012    | 0.000   | 0.120    |
| Nonperforming loans/assets  | 464  | 0.069  | 0.042  | 0.089    | 0.001   | 0.800    |
| Equity/assets               | 464  | 0.100  | 0.049  | 0.159    | 0.000   | 0.995    |
| ln(assets)                  | 464  | 15.261 | 15.019 | 2.148    | 10.390  | 20.596   |
| Conglomerate affiliation    | 464  | 0.317  | 0.000  | 0.466    | 0.000   | 1.000    |
| Hyper-regulated period      | 464  | 0.328  | 0.000  | 0.470    | 0.000   | 1.000    |

#### Table 2 Summary statistics of economies of diversification measures for loan portfolios

Table 2 presents summary statistics of the profit premium and cost discount measured by economies of diversification for loan portfolios. The definitions of the decompositions of the portfolios are the same as in Table 1. Profit premium is defined as the proportional difference in predicted profits between the observed diversified bank and a hypothetical quasi-focused bank with the same total amount of output, while a hypothetical quasi-focused bank is defined in our paper as the hypothetical focused bank which produces at least the minimum in each of the products. Cost discount is defined as the proportional difference in predicted costs between a hypothetical quasi-focused bank and the observed diversified bank.

|   | Obs | Mean   | Median | S.D.   | Minimum | Maximum |
|---|-----|--------|--------|--------|---------|---------|
| Economies of Diversification in Loan Portfolios |     |        |        |        |         |         |
| Profit premium                                  |     |        |        |        |         |         |
| Industry loans                                  | 464 | -0.007 | 0.000  | -0.129 | -0.962  | 0.828   |
| Commercial loans                                | 464 | -0.042 | 0.000  | 0.187  | -0.953  | 0.000   |
| Real estate loans                               | 464 | -0.075 | 0.000  | -0.243 | -1.010  | 0.000   |
| Agriculture loans                               | 464 | -0.582 | -0.823 | -0.403 | -1.100  | 0.000   |
| Consumer loans                                  | 464 | -0.311 | 0.000  | -0.413 | -1.100  | 0.000   |
| Average profit premium for loan portfolios      | 464 | -0.203 | -0.174 | 0.170  | -0.769  | 0.166   |
| Cost discount                                   |     |        |        |        |         |         |
| Industry loans                                  | 464 | -0.029 | -0.027 | 0.057  | -0.555  | 0.840   |
| Commercial loans                                | 464 | -0.028 | -0.025 | -0.058 | -0.904  | 0.676   |
| Real estate loans                               | 464 | -0.249 | -0.216 | 0.572  | -0.602  | 0.834   |
| Agriculture loans                               | 464 | -0.112 | -0.095 | -0.274 | -0.291  | 0.402   |
| Consumer loans                                  | 464 | -0.184 | -0.159 | 0.445  | -0.469  | 0.647   |
| Average cost discount for loan portfolios       | 464 | -0.117 | -0.104 | 0.082  | -0.589  | 0.317   |

Table 3 Summary statistics of economies of diversification measures for loan, deposit, asset, and geographic portfolios

Table 3 presents summary statistics of the economies of diversification measures (i.e., profit premium and cost discount). Diversifications are defined in four dimensions: loans, deposits, assets, and geography. In addition to the presentation of summary statistics of the overall sample Chinese banks, summary statistics of subsamples of Chinese banks by size (i.e., small banks, medium banks, and large banks) are also presented. More specifically, bank size is defined based on total assets (inflation-adjusted to the base year 1996) of the bank at year t, and the bank is a small bank if its assets are less than or equal to US \$ 1 billion, medium bank if the bank's assets are greater than US \$1 billion but less than or equal to US \$20 billion; large bank if the bank's assets are greater than US \$20 billion. The definitions of decompositions of the loan, deposit, asset, and geographic portfolios are the same as in Table 1.

|                     | Diversification Type               | Obs        | Mean   | Median       | Std Dev | Minimu<br>m | Maximum |
|---------------------|------------------------------------|------------|--|--------------|---------|-------------|---------|
| Panel A: All obser  | rvations                           |            |  |              |         |             |         |
|                     | Loan diversification               | 464        | -0.203   | -0.174       | 0.170   | -0.769      | 0.166   |
| Des fit Des estimat | Deposit diversification            | 464        | -0.052   | -0.054       | 0.358   | -1.981      | 2.021   |
| Profit Premium      | Asset diversification              | 464        | -0.159   | -0.151       | 0.031   | -0.318      | -0.107  |
|                     | Geographical diversification       | 464        | -0.049   | -0.050       | 0.021   | -0.160      | 0.015   |
|                     | Loan diversification               | 464        | -0.117   | -0.104       | 0.082   | -0.589      | 0.317   |
| Cent Discount       | Deposit diversification            | 464        | -0.073   | -0.052       | 0.358   | -1.500      | 1.500   |
| Cost Discount       | Asset diversification              | 464        | -0.233   | -0.186       | 0.182   | -1.235      | 1.492   |
|                     | Geographical diversification       | 464        | -0.452   | -0.434       | 0.943   | -2.516      | 1.209   |
| Panel B: Summar     | y statistics for small banks: asse | ts < US    | \$1 billion  |              |         |             |         |
|                     | Loan diversification               | 100        | -0.130   | 0.000        | 0.169   | -0.607      | 0.166   |
| Drofit Dromium      | Deposit diversification            | 100        | -0.014   | -0.020       | 0.197   | -0.355      | 1.529   |
| FIOIII FIEIIIIUIII  | Asset diversification              | 100        | -0.162   | -0.153       | 0.033   | -0.267      | -0.107  |
|                     | Geographical diversification       | 100        | -0.046   | -0.052       | 0.019   | -0.081      | 0.015   |
|                     | Loan diversification               | 100        | -0.090   | -0.078       | 0.087   | -0.589      | 0.010   |
| Cost Discount       | Deposit diversification            | 100        | -0.001   | -0.035       | 0.358   | -1.500      | 1.500   |
|                     | Asset diversification              | 100        | -0.244   | -0.197       | 0.260   | -1.235      | 1.492   |
|                     | Geographical diversification       | 100        | -0.706   | -0.486       | 0.951   | -2.516      | 1.209   |
| Panel C: Summar     | y statistics for medium banks: U   | S \$1 bill | ion < assets < asset as a set of a se | < US \$20 bi | llion   |             |         |
|                     | Loan diversification               | 261        | -0.206   | -0.174       | 0.164   | -0.705      | 0.142   |
| Drofit Dromium      | Deposit diversification            | 261        | -0.054   | -0.054       | 0.420   | -1.981      | 2.021   |
| FIOIII FIEIIIIUIII  | Asset diversification              | 261        | -0.163   | -0.152       | 0.031   | -0.318      | -0.113  |
|                     | Geographical diversification       | 261        | -0.049   | -0.050       | 0.020   | -0.160      | 0.011   |
|                     | Loan diversification               | 261        | -0.127   | -0.105       | 0.080   | -0.589      | 0.317   |
| Cost Discount       | Deposit diversification            | 261        | -0.111   | -0.059       | 0.406   | -1.500      | 1.500   |
| Cost Discoult       | Asset diversification              | 261        | -0.243   | -0.191       | 0.163   | -1.235      | 0.199   |
|                     | Geographical diversification       | 261        | -0.702   | -0.651       | 0.849   | -2.516      | 1.209   |
| Panel D: Summar     | y statistics for large banks: asse | ts > US    | \$20 billion   |              |         |             |         |
|                     | Loan diversification               | 103        | -0.267   | -0.205       | 0.161   | -0.769      | 0.000   |
| Profit Promium      | Deposit diversification            | 103        | -0.083   | -0.069       | 0.303   | -1.981      | 2.021   |
| r tont r tennum     | Asset diversification              | 103        | -0.148   | -0.142       | 0.027   | -0.318      | -0.119  |
|                     | Geographical diversification       | 103        | -0.054   | -0.048       | 0.025   | -0.160      | 0.004   |
|                     | Loan diversification               | 103        | -0.119   | -0.110       | 0.075   | -0.589      | 0.317   |
| Cost Dissourt       | Deposit diversification            | 103        | -0.048   | -0.055       | 0.166   | -0.270      | 1.500   |
| Cost Discount       | Asset diversification              | 103        | -0.196   | -0.167       | 0.123   | -1.235      | -0.129  |
|                     | Geographical diversification       | 103        | 0.429  | 0.558        | 0.563   | -1.127      | 1.209   |

#### Table 4: OLS regressions of profit economies of diversification (profit premium) on firm-level characteristics

Table 4 presents the ordinary least squares (OLS) regressions of profit economies of diversification (i.e., profit premium) on ownership variables, risk proxies, size variables (including both ln(assets) and squared term of ln(assets)), conglomerate affiliation dummy (equal to 1 if the bank is affiliated with a conglomerate, 0 otherwise), and hyper-regulated period dummy (equal to 1 for period 1996-2001, 0 otherwise). Domestic private ownership and the less-regulated period are considered omitted variables in the regressions. The profit premiums in the first, second, third and fourth paired columns are calculated based on decompositions of loan, deposit, asset, and geographic portfolios, respectively, and these decompositions follow the same definitions as described in previous tables. Absolute values of t statistics (based on White heteroskedastic-consistent standard errors) are presented in brackets. \*, \*\*, and \*\*\* represent significance level of 10%, 5%, and 1% respectively.

|                         | Lo           | an           | Dep          | osit         | As            | set           | Geographical  |               |
|-------------------------|--------------|--------------|--------------|--------------|---------------|---------------|---------------|---------------|
|                         | Diversi      | fication     | Diversi      | fication     | Diversi       | fication      | Diversit      | fication      |
| Constant                | $-0.548^{*}$ | -0.474       | -0.486*      | -0.403       | -0.147**      | -0.164**      | -0.282***     | -0.266***     |
|                         | [1.70]       | [1.60]       | [1.73]       | [0.70]       | [1.98]        | [2.07]        | [2.96]        | [2.86]        |
| Big Four                | -0.100       | -0.118*      | -0.168       | -0.191       | -0.034**      | -0.029*       | $0.035^{***}$ | 0.039***      |
|                         | [1.38]       | [1.67]       | [0.87]       | [0.97]       | [2.26]        | [1.93]        | [3.80]        | [3.86]        |
| State ownership         | -0.018       | -0.020       | -0.138**     | -0.135***    | -0.026***     | -0.027***     | -0.007        | -0.007        |
|                         | [0.53]       | [0.58]       | [2.29]       | [2.26]       | [4.03]        | [4.07]        | [1.31]        | [1.45]        |
| Foreign ownership       | $0.016^{**}$ | $0.010^{**}$ | $0.010^{**}$ | $0.014^{**}$ | $0.019^{***}$ | $0.018^{***}$ | 0.004         | 0.005         |
|                         | [2.51]       | [2.32]       | [2.11]       | [2.15]       | [2.98]        | [2.95]        | [0.91]        | [1.23]        |
| Loan loss               | -0.928       |              | -1.702       |              | -0.398**      |               | -0.170        |               |
| provisions/assets       | [1.41]       |              | [1.25]       |              | [2.47]        |               | [1.34]        |               |
| Nonperforming           |              | -0.065       |              | -0.017       |               | -0.002        |               | -0.017        |
| loans/assets            |              | [0.68]       |              | [0.13]       |               | [0.10]        |               | [1.56]        |
| ln(assets)              | 0.067        | 0.060        | -0.186       | -0.178       | -0.041**      | -0.043**      | $0.030^{**}$  | $0.029^{**}$  |
|                         | [0.66]       | [0.58]       | [0.72]       | [0.69]       | [2.16]        | [2.20]        | [2.43]        | [2.33]        |
| Squared ln(assets)      | -0.003       | -0.003       | 0.006        | 0.006        | $0.001^{**}$  | $0.002^{**}$  | -0.001**      | -0.001**      |
|                         | [0.87]       | [0.83]       | [0.68]       | [0.67]       | [2.42]        | [2.42]        | [2.37]        | [2.30]        |
| Equity/assets           | $0.206^{**}$ | $0.166^{*}$  | 0.217        | 0.147        | 0.016         | 0.001         | $0.057^{***}$ | $0.050^{***}$ |
|                         | [2.10]       | [1.74]       | [0.88]       | [0.61]       | [0.78]        | [0.00]        | [3.77]        | [3.95]        |
| Conglomerate affiliated | 0.010        | 0.011        | $0.072^*$    | $0.072^{*}$  | 0.003         | 0.003         | -0.013***     | -0.012***     |
|                         | [0.50]       | [0.57]       | [1.82]       | [1.78]       | [0.78]        | [0.79]        | [4.22]        | [4.10]        |
| Hyper-regulated period  | 0.001        | 0.001        | 0.049        | 0.051        | -0.019***     | -0.019***     | $0.003^{*}$   | $0.004^{*}$   |
|                         | [0.02]       | [0.03]       | [1.25]       | [1.24]       | [6.42]        | [6.43]        | [1.65]        | [1.76]        |
| Observations            | 463          | 463          | 463          | 463          | 463           | 463           | 463           | 463           |
| R-square                | 0.11         | 0.11         | 0.04         | 0.04         | 0.20          | 0.19          | 0.14          | 0.13          |
| Adj R-square            | 0.09         | 0.09         | 0.02         | 0.02         | 0.19          | 0.17          | 0.12          | 0.12          |
| F-statistics            | 11.25        | 12.79        | 3.51         | 4.42         | 14.51         | 13.58         | 6.36          | 6.49          |

#### Table 5: OLS Regressions of cost economies of diversification (cost discount) on firm-level characteristics

Table 5 presents the ordinary least squares (OLS) regressions of cost economies of diversification (cost discount) on ownership variables, risk proxies, size variables (including both ln(assets) and squared term of ln(assets)), conglomerate affiliation dummy (equal to 1 if the bank is affiliated with a conglomerate, 0 otherwise), and hyper-regulated period dummy (equal to 1 for period 1996-2001, 0 otherwise). Domestic private ownership and the less-regulated period are considered omitted variables in the regressions. The cost discounts in the first, second, third and fourth paired columns are calculated based on decompositions of loan, deposit, asset, and geographical portfolios, respectively, and these decompositions follow the same definitions as described in previous tables. Absolute values of t statistics (based on White heteroskedastic-consistent standard errors) are presented in brackets. \*, \*\*, and \*\*\* represent significance level of 10%, 5%, and 1% respectively.

|                             | Lo          | oan           | Dep          | posit        | As           | sset         | Geographical  |               |
|-----------------------------|-------------|---------------|--------------|--------------|--------------|--------------|---------------|---------------|
|                             | Diversi     | fication      | Diversi      | fication     | Diversi      | fication     | Diversi       | fication      |
| Constant                    | -0.098*     | $-0.070^{*}$  | $-0.097^{*}$ | -0.051**     | -0.656*      | -0.732*      | -1.640***     | -1.918***     |
|                             | [1.74]      | [1.73]        | [1.94]       | [1.97]       | [1.67]       | [1.71]       | [2.82]        | [2.88]        |
| Big Four                    | -0.048**    | -0.043*       | $-0.279^{*}$ | -0.263*      | -0.096**     | $-0.068^{*}$ | $0.298^{*}$   | $0.210^{*}$   |
|                             | [2.00]      | [1.88]        | [1.88]       | [1.69]       | [2.07]       | [1.77]       | [1.96]        | [1.69]        |
| State ownership             | -0.045***   | -0.047***     | -0.107       | -0.104       | -0.099*      | -0.105*      | -0.406**      | -0.423**      |
|                             | [2.92]      | [3.03]        | [1.03]       | [0.99]       | [1.71]       | [1.78]       | [2.32]        | [2.40]        |
| Foreign ownership           | 0.033**     | $0.037^{***}$ | $0.014^{**}$ | $0.013^{**}$ | $0.099^{**}$ | 0.103**      | $0.847^{***}$ | $0.848^{***}$ |
|                             | [2.42]      | [2.78]        | [2.15]       | [2.14]       | [2.55]       | [2.42]       | [4.67]        | [4.57]        |
| Loan loss provisions/assets | -1.022***   |               | -1.466       |              | -3.105***    |              | -8.369**      |               |
|                             | [2.91]      |               | [0.91]       |              | [2.73]       |              | [2.10]        |               |
| Nonperforming               |             | -0.084***     |              | -0.019       |              | -0.118       |               | -0.174        |
| loans/assets                |             | [2.76]        |              | [0.14]       |              | [1.16]       |               | [0.29]        |
| ln(assets)                  | -0.100*     | -0.100*       | -0.536**     | -0.541**     | -0.113       | -0.120       | -1.548***     | -1.574***     |
|                             | [1.94]      | [1.92]        | [2.11]       | [2.13]       | [0.88]       | [0.92]       | [3.19]        | [3.24]        |
| Squared ln(assets)          | $0.003^{*}$ | $0.003^{*}$   | $0.018^{**}$ | $0.018^{**}$ | 0.004        | 0.004        | $0.055^{***}$ | $0.055^{***}$ |
|                             | [1.93]      | [1.87]        | [2.22]       | [2.23]       | [0.98]       | [0.97]       | [3.51]        | [3.54]        |
| Equity/assets               | 0.077       | 0.038         | -0.175       | -0.235       | 0.036        | -0.087       | -0.444        | -0.780        |
|                             | [1.46]      | [0.80]        | [0.58]       | [0.79]       | [0.28]       | [0.61]       | [0.88]        | [1.54]        |
| Conglomerate affiliated     | 0.007       | 0.005         | -0.095       | -0.095       | 0.018        | 0.014        | 0.393***      | $0.387^{***}$ |
|                             | [0.72]      | [0.46]        | [1.65]       | [1.62]       | [0.50]       | [0.40]       | [3.78]        | [3.71]        |
| Hyper-regulated period      | -0.043***   | -0.046***     | -0.028       | -0.030       | -0.085***    | -0.092***    | -0.188**      | -0.202**      |
|                             | [6.19]      | [6.45]        | [0.72]       | [0.73]       | [4.35]       | [4.28]       | [2.28]        | [2.32]        |
| Observations                | 463         | 463           | 463          | 463          | 463          | 463          | 463           | 463           |
| R-square                    | 0.21        | 0.20          | 0.04         | 0.03         | 0.12         | 0.10         | 0.32          | 0.32          |
| Adj R-square                | 0.19        | 0.18          | 0.02         | 0.01         | 0.11         | 0.08         | 0.31          | 0.30          |
| F-statistics                | 24.77       | 39.00         | 2.12         | 2.36         | 6.32         | 6.25         | 37.50         | 36.39         |

Table 6 Summary statistics of Focus Indices of loan, deposit, asset, and geographic portfolios

Table 6 presents the summary statistics of Focus Indices, and decompositions of loan, deposit, asset and geographical portfolios follow the same definitions as in Table 1. Summary statistics of subsamples of Chinese banks by size are also presented.

|                                    | Obs.              | Mean             | Median          | Std. Dev | Minimum | Maximum |
|------------------------------------|-------------------|------------------|-----------------|----------|---------|---------|
| Summary Statistics for all observe | ations            |                  |                 |          |         |         |
| Loan Focus Index                   | 464               | 0.428            | 0.415           | 0.086    | 0.282   | 0.700   |
| Deposit Focus Index                | 464               | 0.530            | 0.401           | 0.122    | 0.273   | 1.000   |
| Asset Focus Index                  | 464               | 0.420            | 0.520           | 0.090    | 0.281   | 0.979   |
| Geographic Focus Index             | 464               | 0.866            | 1.000           | 0.286    | 0.216   | 1.000   |
| Summary statistics for small bank  | ks: $assets < US$ | \$1 billion      |                 |          |         |         |
| Loan Focus Index                   | 100               | 0.426            | 0.412           | 0.082    | 0.298   | 0.609   |
| Deposit Focus Index                | 100               | 0.600            | 0.446           | 0.179    | 0.335   | 1.000   |
| Asset Focus Index                  | 100               | 0.484            | 0.568           | 0.135    | 0.307   | 0.979   |
| Geographic Focus Index             | 100               | 1.000            | 1.000           | 0.000    | 1.000   | 1.000   |
| Summary statistics for medium be   | anks: US \$1 bi   | llion < assets < | US \$20 billion | ļ        |         |         |
| Loan Focus Index                   | 261               | 0.411            | 0.397           | 0.075    | 0.282   | 0.610   |
| Deposit Focus Index                | 261               | 0.528            | 0.383           | 0.098    | 0.273   | 0.941   |
| Asset Focus Index                  | 261               | 0.393            | 0.528           | 0.056    | 0.281   | 0.564   |
| Geographic Focus Index             | 261               | 0.970            | 1.000           | 0.142    | 0.270   | 1.000   |
| Summary statistics for large bank  | ks: assets > US   | S \$20 billion   |                 |          |         |         |
| Loan Focus Index                   | 103               | 0.474            | 0.471           | 0.098    | 0.299   | 0.700   |
| Deposit Focus Index                | 103               | 0.470            | 0.428           | 0.061    | 0.336   | 0.593   |
| Asset Focus Index                  | 103               | 0.428            | 0.466           | 0.070    | 0.299   | 0.644   |
| Geographic Focus Index             | 103               | 0.470            | 1.000           | 0.342    | 0.216   | 1.000   |

#### Table 7 OLS regressions of ROA on Focus Indices and ownership.

Table 7 presents the ordinary least squares (OLS) regressions of return on assets (ROA) on Focus Indices, ownership variables, conglomerate affiliation dummy (equal to 1 if the bank is affiliated with a conglomerate, 0 otherwise), the interaction terms between Focus Indices and ownership variables, between Focus Indices and conglomerate affiliation dummy, and control variables including risk proxies, size variables (including both ln(assets) and squared term of ln(assets)), and hyper-regulated period dummy (equal to 1 for period 1996-2001, 0 otherwise). Domestic private ownership and the less-regulated period are considered omitted variables in the regressions. The Focus Index in the first paired columns represents the loan Focus Index, and the same Focus Index is also used to construct the interaction terms between Focus Indices and other variables. The Focus Indices in the second, third, and four paired columns represent deposit Focus Index, asset Focus Index, geographic Focus Index. Absolute values of t statistics (based on White heteroskedastic-consistent standard errors) are presented in brackets. \*, \*\*, and \*\*\* represent significance level of 10%, 5%, and 1% respectively.

|                            | Dependent Variable: ROA |              |               |               |              |             |               |              |  |
|----------------------------|-------------------------|--------------|---------------|---------------|--------------|-------------|---------------|--------------|--|
|                            | Lo                      | an           | Dep           | osit          | A            | sset        | Geogra        | aphical      |  |
|                            | Diversi                 | fication     | Diversi       | fication      | Divers       | ification   | Diversit      | fication     |  |
| Constant                   | -0.024                  | -0.019       | -0.021        | -0.020        | -0.019       | -0.011      | -0.029        | -0.023       |  |
|                            | [0.45]                  | [0.36]       | [0.44]        | [0.43]        | [0.37]       | [0.22]      | [0.52]        | [0.42]       |  |
| Big Four                   | -0.010                  | -0.008       | -0.008        | -0.008        | 0.005        | 0.003       | -0.085***     | -0.072***    |  |
|                            | [1.56]                  | [1.36]       | [0.98]        | [0.97]        | [0.78]       | [0.58]      | [4.24]        | [4.16]       |  |
| State ownership            | 0.002                   | 0.001        | 0.005         | 0.005         | 0.001        | 0.001       | -0.001        | -0.001       |  |
|                            | [0.28]                  | [0.07]       | [0.58]        | [0.60]        | [0.21]       | [0.12]      | [0.88]        | [0.57]       |  |
| Foreign ownership          | $0.004^{**}$            | $0.004^{**}$ | $0.010^{*}$   | $0.009^{**}$  | $0.005^{*}$  | 0.003       | 0.012         | 0.011        |  |
|                            | [2.35]                  | [2.32]       | [1.93]        | [1.98]        | [1.86]       | [1.28]      | [1.40]        | [1.15]       |  |
| Focus Index                | $0.004^{**}$            | $0.002^{**}$ | $0.007^{***}$ | $0.006^{***}$ | $0.002^{**}$ | $0.007^{*}$ | 0.002         | 0.002        |  |
|                            | [2.39]                  | [2.16]       | [2.62]        | [2.60]        | [2.32]       | [1.84]      | [0.77]        | [0.68]       |  |
| Focus Index $\times$ Big   | $0.021^{*}$             | $0.020^{*}$  | 0.017         | 0.019         | -0.011       | -0.005      | $0.376^{***}$ | 0.321***     |  |
| Four                       | [1.72]                  | [1.68]       | [0.89]        | [0.95]        | [0.76]       | [0.35]      | [4.03]        | [4.06]       |  |
| Focus Index $\times$ State | -0.007                  | -0.003       | -0.012        | -0.013        | -0.004       | -0.003      | -0.001        | -0.001       |  |
| ownership                  | [0.35]                  | [0.14]       | [0.63]        | [0.64]        | [0.31]       | [0.20]      | [0.20]        | [0.10]       |  |
| Focus Index × Foreign      | -0.008**                | -0.009**     | $-0.007^{*}$  | -0.003**      | -0.005***    | -0.011***   | -0.005**      | -0.004**     |  |
| ownership                  | [2.31]                  | [2.39]       | [1.69]        | [2.30]        | [2.31]       | [2.63]      | [2.57]        | [2.38]       |  |
| Focus Index $\times$       | -0.006*                 | $-0.007^{*}$ | -0.006*       | -0.006*       | -0.001**     | -0.004      | -0.001*       | $-0.002^{*}$ |  |
| Conglomerate               |                         |              |               |               |              |             |               |              |  |
| affiliated                 | [1.70]                  | [1.86]       | [1.90]        | [1.85]        | [2.10]       | [1.51]      | [1.66]        | [1.78]       |  |
| Loan loss                  | -0.032                  |              | -0.029        |               | -0.034       |             | -0.044        |              |  |
| provisions/assets          | [0.72]                  |              | [0.67]        |               | [0.65]       | *           | [0.94]        |              |  |
| Nonperforming              |                         | -0.005       |               | -0.005        |              | -0.007*     |               | -0.004       |  |
| loans/assets               |                         | [1.58]       |               | [0.96]        |              | [1.95]      |               | [1.10]       |  |
| ln(assets)                 | 0.005                   | 0.004        | 0.004         | 0.004         | 0.004        | 0.003       | 0.006         | 0.005        |  |
|                            | [0.67]                  | [0.62]       | [0.66]        | [0.68]        | [0.64]       | [0.54]      | [0.77]        | [0.69]       |  |
| Squared ln(assets)         | 0.001                   | 0.001        | 0.001         | 0.001         | 0.001        | 0.001       | 0.001         | 0.001        |  |
|                            | [0.66]                  | [0.62]       | [0.65]        | [0.68]        | [0.62]       | [0.53]      | [0.75]        | [0.69]       |  |
| Equity/assets              | 0.016                   | 0.014        | 0.017         | 0.016         | 0.016        | 0.015       | 0.018         | 0.015        |  |
|                            | [1.53]                  | [1.61]       | [1.17]        | [1.21]        | [1.32]       | [1.43]      | [1.58]        | [1.60]       |  |
| Conglomerate               | 0.002                   | 0.003        | 0.003         | 0.003         | -0.001       | -0.002      | -0.002        | -0.002       |  |
| affiliated                 | [0.70]                  | [0.88]       | [0.39]        | [0.41]        | [0.13]       | [0.47]      | [1.08]        | [1.14]       |  |
| Hyper-regulated period     | 0.001                   | 0.001        | 0.001         | 0.001         | 0.001        | 0.001       | 0.001         | 0.001        |  |
|                            | [1.30]                  | [1.34]       | [1.23]        | [1.19]        | [0.91]       | [1.04]      | [1.42]        | [1.39]       |  |
| Observations               | 463                     | 463          | 463           | 463           | 463          | 463         | 463           | 463          |  |
| R-square                   | 0.24                    | 0.24         | 0.24          | 0.24          | 0.24         | 0.24        | 0.25          | 0.24         |  |
| Adj R-square               | 0.21                    | 0.22         | 0.21          | 0.21          | 0.21         | 0.22        | 0.22          | 0.22         |  |
| F-statistics               | 3.43                    | 3.47         | 4.13          | 4.78          | 4.19         | 4.83        | 11.10         | 12.67        |  |

#### Table 8 OLS regressions of total expenses/total assets on Focus Indices and ownership

Table 8 presents the ordinary least squares (OLS) regressions of total expenses/total assets on Focus Indices, ownership variables, conglomerate affiliation dummy (equal to 1 if the bank is affiliated with a conglomerate, 0 otherwise), the interaction terms between Focus Indices and ownership variables, between Focus Indices and conglomerate affiliation dummy, and control variables including risk proxies, size variables (including both ln(assets) and squared term of ln(assets)), and hyper-regulated period dummy (equal to 1 for period 1996-2001, 0 otherwise). Domestic private ownership and the less-regulated period are considered omitted variables in the regressions. The Focus Index in the first paired columns represents the loan Focus Index, and the same Focus Index is also used to construct the interaction terms between Focus Indices and other variables. The Focus Indices in the second, third, and four paired columns represent deposit Focus Index, asset Focus Index, geographic Focus Index. Absolute values of t statistics (based on White heteroskedastic-consistent standard errors) are presented in brackets. \*, \*\*\*, and \*\*\*\* represent significance level of 10%, 5%, and 1% respectively.

|                         |               | Dependent Variable: Total Expenses/Total Assets |               |                |               |               |               |               |  |  |  |
|-------------------------|---------------|---|---------------|----------------|---------------|---------------|---------------|---------------|--|--|--|
|                         | Lo            | an  | Dep           | osit           | As            | set           | Geog          | raphical      |  |  |  |
|                         | Diversi       | fication  | Diversi       | fication       | Diversi       | fication      | Divers        | sification    |  |  |  |
| Constant                | -0.147***     | -0.151***                                       | -0.136**      | -0.137**       | -0.167***     | -0.178***     | -0.125**      | -0.130**      |  |  |  |
|                         | [2.76]        | [2.81]  | [2.49]        | [2.51]         | [3.25]        | [3.38]        | [2.48]        | [2.54]        |  |  |  |
| Big Four                | 0.001         | 0.001   | $0.039^{***}$ | $0.043^{***}$  | $0.028^{**}$  | $0.030^{**}$  | 0.011         | 0.001         |  |  |  |
| -                       | [0.05]        | [0.05]  | [3.01]        | [3.38]         | [2.29]        | [2.41]        | [0.27]        | [0.03]        |  |  |  |
| State ownership         | 0.027         | 0.028   | -0.031**      | -0.035**       | 0.006         | 0.006         | -0.004        | -0.005        |  |  |  |
| -                       | [1.54]        | [1.61]  | [2.09]        | [2.40]         | [0.46]        | [0.45]        | [1.10]        | [1.37]        |  |  |  |
| Foreign ownership       | $0.027^{*}$   | $0.029^{**}$                                    | $0.001^{**}$  | $0.003^{**}$   | $0.023^{**}$  | $0.025^{**}$  | $0.002^{**}$  | $0.005^{**}$  |  |  |  |
|                         | [1.95]        | [2.06]  | [2.08]        | [2.34]         | [2.02]        | [2.22]        | [2.14]        | [2.31]        |  |  |  |
| Focus Index             | -0.012        | -0.014  | -0.025        | $-0.029^{*}$   | 0.039**       | 0.043**       | -0.003        | -0.002        |  |  |  |
|                         | [0.59]        | [0.68]  | [1.50]        | [1.80]         | [2.09]        | [2.27]        | [0.48]        | [0.32]        |  |  |  |
| Focus Index × Big Four  | -0.014        | -0.013  | -0.073***     | $-0.079^{***}$ | $-0.054^{*}$  | -0.062**      | -0.001        | -0.050        |  |  |  |
| -                       | [0.61]        | [0.56]  | [2.75]        | [3.14]         | [1.91]        | [2.13]        | [0.01]        | [0.30]        |  |  |  |
| Focus Index × State     | -0.039        | -0.042  | $0.087^{***}$ | $0.095^{***}$  | -0.009        | -0.009        | -0.021***     | -0.022***     |  |  |  |
| ownership               | [1.02]        | [1.10]  | [2.86]        | [3.21]         | [0.32]        | [0.30]        | [3.08]        | [3.30]        |  |  |  |
| Focus Index × Foreign   | $0.052^{*}$   | $0.056^{*}$                                     | $0.017^{**}$  | $0.025^{**}$   | $0.039^{*}$   | $0.044^{*}$   | $0.005^{**}$  | $0.003^{**}$  |  |  |  |
| ownership               | [1.66]        | [1.77]  | [2.10]        | [2.57]         | [1.67]        | [1.94]        | [2.30]        | [2.19]        |  |  |  |
| Focus Index ×           | $0.010^{*}$   | $0.012^{*}$                                     | $0.021^{*}$   | $0.022^{**}$   | $0.010^{*}$   | $0.015^{*}$   | $0.007^{**}$  | $0.007^{**}$  |  |  |  |
| Conglomerate affiliated | [1.72]        | [1.85]  | [1.84]        | [2.02]         | [1.61]        | [1.92]        | [2.32]        | [2.23]        |  |  |  |
| Loan loss               | 0.085         |   | 0.054         |                | 0.070         |               | 0.073         |               |  |  |  |
| provisions/assets       | [1.43]        |   | [0.95]        |                | [1.21]        |               | [1.26]        |               |  |  |  |
| Nonperforming           |               | 0.002   |               | 0.009          |               | 0.007         |               | 0.001         |  |  |  |
| loans/assets            |               | [0.39]  |               | [1.26]         |               | [1.27]        |               | [0.01]        |  |  |  |
| ln(assets)              | $0.021^{***}$ | $0.021^{***}$                                   | $0.021^{***}$ | $0.022^{***}$  | $0.022^{***}$ | $0.023^{***}$ | $0.018^{***}$ | $0.018^{***}$ |  |  |  |
|                         | [3.10]        | [3.13]  | [3.00]        | [3.07]         | [3.38]        | [3.46]        | [2.67]        | [2.75]        |  |  |  |
| Squared ln(assets)      | -0.001***     | -0.001***                                       | -0.001***     | -0.001***      | -0.001***     | -0.001***     | -0.001***     | -0.001***     |  |  |  |
|                         | [3.18]        | [3.19]  | [3.04]        | [3.10]         | [3.46]        | [3.52]        | [2.68]        | [2.74]        |  |  |  |
| Equity/assets           | 0.006         | 0.009   | -0.004        | -0.002         | -0.002        | 0.002         | 0.003         | 0.007         |  |  |  |
|                         | [0.79]        | [1.30]  | [0.48]        | [0.25]         | [0.24]        | [0.20]        | [0.43]        | [0.87]        |  |  |  |
| Conglomerate affiliated | -0.008        | -0.008  | -0.013**      | -0.013**       | 0.001         | 0.002         | 0.005         | 0.004         |  |  |  |
|                         | [1.13]        | [1.24]  | [2.23]        | [2.36]         | [0.03]        | [0.31]        | [1.06]        | [0.99]        |  |  |  |
| Hyper-regulated period  | $0.011^{***}$ | $0.011^{***}$                                   | $0.012^{***}$ | $0.012^{***}$  | $0.012^{***}$ | $0.012^{***}$ | $0.011^{***}$ | $0.011^{***}$ |  |  |  |
|                         | [7.61]        | [7.62]  | [8.53]        | [8.51]         | [8.59]        | [8.39]        | [7.81]        | [7.64]        |  |  |  |
| Observations            | 463           | 463   | 463           | 463            | 463           | 463           | 463           | 463           |  |  |  |
| R-square                | 0.27          | 0.26  | 0.29          | 0.29           | 0.29          | 0.29          | 0.28          | 0.28          |  |  |  |
| Adj R-square            | 0.24          | 0.24  | 0.27          | 0.27           | 0.27          | 0.27          | 0.26          | 0.26          |  |  |  |
| F-statistics            | 13.43         | 13.31   | 14.17         | 14.01          | 13.17         | 12.75         | 14.22         | 14.32         |  |  |  |

#### Table 9 OLS regressions of Risk on Focus Indices and ownership

Table 9 presents the ordinary least squares (OLS) regressions of loan loss provisions/assets (the left four columns) and nonperforming loans/assets (the right four columns) on Focus Indices, ownership variables, conglomerate affiliation dummy (equal to 1 if the bank is affiliated with a conglomerate, 0 otherwise), the interaction terms between Focus Indices and ownership variables, between Focus Indices and conglomerate affiliation dummy, and control variables including equity ratio, size variables (including both ln(assets) and squared term of ln(assets)), and hyper-regulated period dummy (equal to 1 for period 1996-2001, 0 otherwise). Domestic private ownership and the less-regulated period are considered omitted variables in the regressions. The Focus Index in the first column represents the loan Focus Index, and the same Focus Index is used to construct the interaction terms between Focus Indices and other variables. The Focus Index in the second, third, and fourth columns represent deposit Focus Index, asset Focus Index, geographic Focus Index. Absolute values of t statistics (based on White heteroskedastic-consistent standard errors) are presented in brackets. \*, \*\*, and \*\*\*\* represent significance level of 10%, 5%, and 1% respectively.

|                               | Dep             | endent Variable: L | oan Loss Provision | ns/Assets       | Depend          | Dependent Variable: Nonperforming Loans/Assets |                 |                 |  |  |
|-------------------------------|-----------------|--------------------|--------------------|-----------------|-----------------|--|-----------------|-----------------|--|--|
|                               | Loan            | Deposit            | Asset              | Geography       | Loan            | Deposit  | Asset           | Geography       |  |  |
|                               | Diversification | Diversification    | Diversification    | Diversification | Diversification | Diversification                                | Diversification | Diversification |  |  |
| Constant                      | -0.064          | -0.026             | -0.062             | -0.065          | 0.625           | 0.008  | 0.889           | 0.850           |  |  |
|                               | [1.35]          | [0.54]             | [1.23]             | [1.27]          | [1.89]          | [0.03]   | [2.57]          | [2.48]          |  |  |
| Big Four                      | 0.007           | $0.035^{*}$        | 0.017              | $0.160^{***}$   | 0.231**         | 0.165  | 0.148           | 1.596***        |  |  |
|                               | [0.46]          | [1.95]             | [0.97]             | [3.34]          | [2.24]          | [1.39]   | [1.31]          | [5.39]          |  |  |
| State ownership               | 0.022           | 0.039**            | 0.016              | 0.011           | 0.183**         | 0.193***                                       | $0.146^{**}$    | 0.024           |  |  |
|                               | [1.34]          | [2.00]             | [1.03]             | [1.64]          | [2.58]          | [2.59]   | [2.17]          | [1.57]          |  |  |
| Foreign ownership             | $0.019^{*}$     | $0.007^{*}$        | $0.011^{*}$        | 0.032           | 0.051           | $0.410^{***}$                                  | $0.355^{**}$    | 0.056           |  |  |
|                               | [1.86]          | [1.93]             | [1.82]             | [1.74]          | [2.39]          | [7.66]   | [2.57]          | [2.61]          |  |  |
| Focus Index                   | -0.033          | -0.037*            | -0.012             | -0.015*         | -0.258**        | -0.244**                                       | -0.531***       | -0.032          |  |  |
|                               | [1.57]          | [1.76]             | [0.65]             | [1.90]          | [2.34]          | [2.49]   | [4.67]          | [0.89]          |  |  |
| Focus Index × Big Four        | -0.009          | -0.099***          | -0.064*            | -0.679***       | -0.284          | -0.075   | -0.494*         | -6.751***       |  |  |
| -                             | [0.27]          | [2.88]             | [1.82]             | [3.18]          | [1.47]          | [0.31]   | [1.82]          | [5.26]          |  |  |
| Focus Index × State ownership | 0.050           | $0.085^{**}$       | 0.038              | $0.016^{**}$    | $0.441^{***}$   | 0.396**  | $0.388^{**}$    | -0.024          |  |  |
| -                             | [1.26]          | [2.20]             | [1.07]             | [2.00]          | [2.71]          | [2.56]   | [2.39]          | [0.88]          |  |  |
| Focus Index × Foreign         | 0.050           | 0.016              | 0.017              | 0.032           | 0.062           | 0.861***                                       | 0.920***        | 0.010           |  |  |
| ownership                     | [1.60]          | [0.78]             | [0.66]             | [0.75]          | [0.20]          | [9.08]   | [3.11]          | [0.10]          |  |  |
| Focus Index × Conglomerate    | 0.025           | $0.022^{*}$        | 0.060***           | 0.005           | 0.081           | 0.051  | 0.162           | 0.093***        |  |  |
| affiliated                    | [1.33]          | [1.66]             | [4.16]             | [0.92]          | [0.77]          | [0.56]   | [1.62]          | [2.79]          |  |  |
| ln(assets)                    | 0.005           | 0.004              | 0.006              | 0.009           | -0.050          | 0.032  | -0.074*         | -0.089**        |  |  |
|                               | [0.80]          | [0.64]             | [0.92]             | [1.28]          | [1.17]          | [0.86]   | [1.71]          | [1.97]          |  |  |
| Squared ln(assets)            | 0.001           | 0.001              | 0.001              | 0.001           | 0.001           | -0.001   | 0.002           | 0.002*          |  |  |
|                               | [0.35]          | [0.19]             | [0.49]             | [0.90]          | [0.88]          | [1.14]   | [1.51]          | [1.71]          |  |  |
| Equity/assets                 | 0.042***        | 0.034***           | 0.050***           | 0.045***        | -0.040          | 0.032  | 0.015           | -0.096*         |  |  |
| 1 2                           | [4.34]          | [3.22]             | [5.23]             | [4.44]          | [0.85]          | [0.65]   | [0.28]          | [1.89]          |  |  |
| Conglomerate affiliated       | -0.010          | -0.010             | 0.025***           | -0.006          | 0.059           | -0.003   | -0.043          | -0.040          |  |  |
| 6                             | [1.27]          | [1.49]             | [3.82]             | [1.16]          | [1.30]          | [0.07]   | [1.05]          | [1.31]          |  |  |
| Hyper-regulated period        | 0.001           | 0.001              | 0.000              | 0.001           | 0.026***        | 0.032***                                       | 0.024***        | 0.027***        |  |  |
|                               | [1.11]          | [0.98]             | [0.26]             | [0.89]          | [2.81]          | [4.25]   | [2.95]          | [3.10]          |  |  |
| Observations                  | 463             | 463                | 463                | 463             | 463             | 463  | 463             | 463             |  |  |
| R-square                      | 0.21            | 0.22               | 0.25               | 0.22            | 0.24            | 0.41   | 0.31            | 0.28            |  |  |
| Adj R-square                  | 0.19            | 0.20               | 0.23               | 0.20            | 0.22            | 0.40   | 0.29            | 0.26            |  |  |
| F-statistics                  | 5.17            | 7.00               | 8.20               | 8.19            | 9.49            | 19.98  | 11.99           | 37.18           |  |  |

#### Table 10 Summary statistics of variables used in efficiency estimations

Table 10 presents the summary statistics of basic variables used in the profit and cost efficiency estimations. In the translog based estimations of profit (cost) efficiency, output variables considered are total loans, total deposits, liquid assets, and other earning assets, and the input variables are:  $w_1$  (price of funds, proxied by the ratio of interest expenses to total deposits),  $w_2$  (price of fixed capital, proxied by the ratio of other operating expenses to fixed assets),  $w_3$  (price of labor, proxied by the ratio of personnel expenses to total number of employees). The outputs are normalized by total assets. All financial values are inflation-adjusted to the base year 1996.

|  | Obs | Mean   | Median | Std. Dev. | Minimum | Maximum |
|--|-----|--------|--------|-----------|---------|---------|
| Profit (Cost) (in billion US\$)          |     |        |        |           |         |         |
| Total profits                            | 464 | 0.189  | 0.015  | 0.729     | -0.080  | 5.784   |
| Total costs                              | 464 | 0.976  | 0.092  | 2.679     | 0.000   | 21.200  |
| Output Quantities (in billion US\$)      |     |        |        |           |         |         |
| Total loans (y <sub>1</sub> )            | 464 | 23.800 | 1.617  | 67.600    | 0.000   | 426.000 |
| Total deposits $(y_2)$                   | 464 | 39.300 | 2.562  | 114.000   | 0.000   | 791.000 |
| Liquid assets (y <sub>3</sub> )          | 464 | 5.323  | 0.347  | 16.100    | 0.012   | 112.000 |
| Other earning assets (y <sub>4</sub> )   | 464 | 16.900 | 1.356  | 50.700    | 0.005   | 444.000 |
| Input Prices                             |     |        |        |           |         |         |
| Price of funds $(w_1)$                   | 464 | 0.027  | 0.020  | 0.016     | 0.003   | 0.067   |
| Price of fixed capital (w <sub>2</sub> ) | 464 | 0.943  | 0.760  | 0.592     | 0.174   | 2.500   |
| Price of labor (w <sub>3</sub> )         | 464 | 14.418 | 14.528 | 2.876     | 0.720   | 21.105  |
| Total assets (in billon US\$)            | 464 | 44.000 | 3.332  | 125.000   | 0.033   | 880.000 |

#### Table 11 OLS regressions of profit efficiency on Focus Indices and ownership

Table 11 presents the ordinary least squares (OLS) regressions of profit efficiency on Focus Indices, ownership variables, conglomerate affiliation dummy (equal to 1 if the bank is affiliated with a conglomerate, 0 otherwise), the interaction terms between Focus Indices and ownership variables, between Focus Indices and conglomerate affiliation dummy, and control variables including size (including both ln(assets) and squared term of ln(assets)), and hyper-regulated period dummy (equal to 1 for period 1996-2001, 0 otherwise). Domestic private ownership and the less-regulated period are considered omitted variables in the regressions. The Focus Index in the first paired columns represents the loan Focus Index, and the same Focus Index is also used to construct the interaction terms between Focus Indices and other variables. The Focus Indices in the second, third, and fourth paired columns represent deposit Focus Index, asset Focus Index, geographic Focus Index. Absolute values of t statistics (based on White heteroskedastic-consistent standard errors) are presented in brackets. <sup>\*</sup>, <sup>\*\*</sup>, and <sup>\*\*\*</sup> represent significance level of 10%, 5%, and 1% respectively.

|                         | Dependent Variable: Profit Efficiency |               |               |               |               |               |               |               |  |
|-------------------------|---------------------------------------|---------------|---------------|---------------|---------------|---------------|---------------|---------------|--|
|                         | L                                     | oan           | De            | eposit        | А             | sset          | Geog          | raphical      |  |
|                         | Divers                                | ification     | Divers        | sification    | Divers        | ification     | Divers        | ification     |  |
| Constant                | $1.441^{***}$                         | $1.476^{***}$ | $1.515^{***}$ | $1.510^{***}$ | $1.551^{***}$ | $1.612^{***}$ | $1.334^{***}$ | $1.372^{***}$ |  |
|                         | [4.54]                                | [4.79]        | [3.82]        | [3.79]        | [4.60]        | [4.99]        | [3.23]        | [3.41]        |  |
| Big Four                | $-0.082^{*}$                          | $-0.067^{*}$  | -0.002**      | -0.019**      | -0.119**      | -0.128**      | -0.132**      | -0.062**      |  |
|                         | [1.84]                                | [1.67]        | [2.02]        | [2.16]        | [2.39]        | [2.57]        | [2.44]        | [2.21]        |  |
| State ownership         | -0.067                                | -0.057        | -0.019        | -0.006        | -0.217***     | -0.199**      | -0.017        | -0.017        |  |
|                         | [0.45]                                | [0.36]        | [0.13]        | [0.04]        | [2.78]        | [2.55]        | [0.96]        | [0.96]        |  |
| Foreign ownership       | $0.071^{*}$                           | $0.076^{*}$   | $0.029^*$     | 0.013**       | $0.083^{**}$  | $0.047^*$     | $0.035^{**}$  | 0.039**       |  |
|                         | [1.86]                                | [1.89]        | [1.85]        | [2.35]        | [2.45]        | [1.78]        | [2.41]        | [2.50]        |  |
| Focus Index             | $0.068^{**}$                          | $0.054^{**}$  | $0.047^{**}$  | $0.076^{**}$  | 0.091**       | $0.047^{**}$  | $0.046^{**}$  | $0.044^{**}$  |  |
|                         | [2.33]                                | [2.25]        | [2.26]        | [2.42]        | [2.03]        | [2.48]        | [2.36]        | [2.27]        |  |
| Focus Index × Big Four  | 0.193                                 | 0.173         | 0.023         | 0.002         | 0.306*        | 0.331**       | 0.660         | 0.365         |  |
|                         | [1.10]                                | [0.98]        | [0.08]        | [0.01]        | [1.82]        | [2.09]        | [0.50]        | [0.28]        |  |
| Focus Index × State     | -0.144                                | -0.120        | -0.020        | -0.031        | -0.491***     | -0.444***     | -0.017        | -0.018        |  |
| ownership               | [0.43]                                | [0.34]        | [0.06]        | [0.09]        | [3.15]        | [2.84]        | [0.48]        | [0.45]        |  |
| Focus Index × Foreign   | -0.105**                              | -0.106**      | -0.001**      | -0.086**      | -0.128**      | -0.040**      | -0.005**      | -0.007**      |  |
| ownership               | [2.55]                                | [2.54]        | [2.01]        | [2.12]        | [2.20]        | [2.33]        | [2.06]        | [2.08]        |  |
| Focus Index $\times$    | -0.008**                              | -0.006**      | -0.080**      | -0.088**      | -0.093**      | -0.089**      | -0.071**      | -0.066**      |  |
| Conglomerate affiliated | [2.11]                                | [2.07]        | [2.51]        | [2.56]        | [2.32]        | [2.51]        | [2.33]        | [2.18]        |  |
| Loan loss               | -0.104                                |               | -0.141        |               | -0.309        |               | -0.057        |               |  |
| provisions/assets       | [0.27]                                |               | [0.39]        |               | [0.73]        |               | [0.15]        |               |  |
| Nonperforming           |                                       | -0.066**      |               | -0.099***     |               | -0.090***     |               | $-0.050^{*}$  |  |
| loans/assets            |                                       | [2.25]        |               | [3.19]        |               | [3.04]        |               | [1.80]        |  |
| ln(assets)              | -0.110*                               | -0.108*       | -0.125        | -0.117        | -0.122        | -0.105        | -0.083        | -0.085        |  |
|                         | [1.73]                                | [1.90]        | [1.26]        | [1.21]        | [1.46]        | [1.45]        | [1.05]        | [1.21]        |  |
| Squared ln(assets)      | -0.078                                | -0.081*       | -0.081*       | -0.077*       | -0.094**      | -0.099**      | -0.066        | -0.069        |  |
| • • •                   | [1.64]                                | [1.72]        | [1.88]        | [1.78]        | [2.05]        | [2.21]        | [1.21]        | [1.31]        |  |
| Equity/assets           | 0.002                                 | 0.002         | 0.002*        | 0.002         | 0.003*        | 0.003**       | 0.002         | 0.002         |  |
|                         | [1.53]                                | [1.59]        | [1.77]        | [1.64]        | [1.96]        | [2.10]        | [1.10]        | [1.19]        |  |
| Conglomerate affiliated | 0.008                                 | 0.010         | 0.031         | 0.033         | 0.031         | 0.027         | 0.066**       | 0.063**       |  |
| -                       | [0.22]                                | [0.31]        | [0.44]        | [0.45]        | [1.05]        | [1.09]        | [2.29]        | [2.24]        |  |
| Hyper-regulated period  | 0.001                                 | 0.002         | -0.001        | 0.002         | -0.001        | 0.001         | 0.001         | 0.002         |  |
|                         | [0.08]                                | [0.39]        | [0.19]        | [0.37]        | [0.14]        | [0.20]        | [0.03]        | [0.26]        |  |
| Observations            | 463                                   | 463           | 463           | 463           | 463           | 463           | 463           | 463           |  |
| R-square                | 0.08                                  | 0.09          | 0.08          | 0.10          | 0.11          | 0.13          | 0.09          | 0.10          |  |
| Adj R-square            | 0.05                                  | 0.06          | 0.05          | 0.07          | 0.09          | 0.10          | 0.06          | 0.07          |  |
| F-statistics            | 3.46                                  | 3.75          | 3.32          | 3.80          | 5.26          | 5.58          | 5.39          | 5.66          |  |

#### Table 12 OLS regressions of cost efficiency on Focus Indices and ownership

Table 12 presents the ordinary least squares (OLS) regressions of cost efficiency on Focus Indices, ownership variables, conglomerate affiliation dummy (equal to 1 if the bank is affiliated with a conglomerate, 0 otherwise), the interaction terms between Focus Indices and ownership variables, between Focus Indices and conglomerate affiliation dummy, and control variables including size (including both ln(assets) and squared term of ln(assets)), and hyper-regulated period dummy (equal to 1 for period 1996-2001, 0 otherwise). Majority Domestic private ownership and the less-regulated period are considered omitted variables in the regressions. The Focus Index in the first paired columns represents the loan Focus Index, and the same Focus Index is also used to construct the interaction terms between Focus Indices and other variables. The Focus Indices in the second, third, and fourth paired columns represent deposit Focus Index, asset Focus Index, geographic Focus Index. Absolute values of t statistics (based on White heteroskedastic-consistent standard errors) are presented in brackets. \*, \*\*, and \*\*\* represent significance level of 10%, 5%, and 1% respectively.

|                              |              |               | Depe         | ndent Varial | ole: Cost Eff | iciency      |             |               |
|------------------------------|--------------|---------------|--------------|--------------|---------------|--------------|-------------|---------------|
|                              | Lo           | an            | Depo         | osit         | As            | set          | Geog        | graphical     |
|                              | Diversi      | fication      | Diversif     | ication      | Diversit      | fication     | Diver       | sification    |
| Constant                     | -0.766       | -0.655        | -0.692       | -0.680       | -0.598        | -0.416       | -0.521      | -0.360        |
|                              | [1.21]       | [0.99]        | [1.12]       | [1.10]       | [1.02]        | [0.68]       | [0.84]      | [0.55]        |
| Big Four                     | -0.026**     | -0.056**      | -0.083***    | -0.077**     | -0.130**      | $-0.097^{*}$ | 0.683**     | $1.005^{***}$ |
|                              | [2.22]       | [2.49]        | [2.60]       | [2.56]       | [2.06]        | [1.80]       | [2.02]      | [2.93]        |
| State ownership              | -0.081**     | -0.046**      | -0.022**     | -0.030***    | -0.263*       | $-0.280^{*}$ | -0.050      | -0.041        |
|                              | [2.43]       | [2.25]        | [2.12]       | [2.16]       | [1.75]        | [1.93]       | [1.56]      | [1.23]        |
| Foreign ownership            | -0.040**     | -0.047**      | -0.004**     | -0.051**     | -0.091*       | -0.146**     | 0.168       | 0.159         |
|                              | [2.21]       | [2.25]        | [2.04]       | [2.40]       | [1.67]        | [2.04]       | [1.32]      | [1.36]        |
| Focus Index                  | $0.263^{**}$ | $0.212^{*}$   | $0.041^{**}$ | $0.057^{**}$ | 0.287         | $0.383^{**}$ | $0.086^{*}$ | $0.089^{*}$   |
|                              | [2.10]       | [1.90]        | [2.20]       | [2.27]       | [1.53]        | [2.10]       | [1.67]      | [1.67]        |
| Focus Index × Big Four       | -0.031       | -0.056        | 0.245        | 0.283        | -0.277        | -0.161       | -2.804*     | -4.169***     |
|                              | [0.13]       | [0.23]        | [0.86]       | [1.00]       | [1.03]        | [0.62]       | [1.87]      | [2.70]        |
| Focus Index $\times$ State   | -0.222       | -0.140        | 0.011        | 0.026        | $0.621^{*}$   | $0.668^{*}$  | 0.060       | 0.048         |
| ownership                    | [0.49]       | [0.32]        | [0.03]       | [0.07]       | [1.70]        | [1.88]       | [1.07]      | [0.82]        |
| Focus Index $\times$ Foreign | -0.204**     | -0.244**      | -0.057**     | -0.172*      | -0.325**      | -0.472       | -0.120*     | -0.101*       |
| ownership                    | [2.46]       | [2.55]        | [2.30]       | [1.78]       | [2.10]        | [1.53]       | [1.87]      | [1.79]        |
| Focus Index $\times$         | -0.084**     | $-0.058^{**}$ | -0.108       | -0.112*      | $-0.148^{*}$  | -0.206       | -0.049**    | -0.038*       |
| Conglomerate affiliated      | [2.54]       | [2.37]        | [1.64]       | [1.66]       | [1.84]        | [1.23]       | [2.11]      | [1.84]        |
| Loan loss                    | -0.680       |               | -0.498       |              | -0.506        |              | -0.550      |               |
| provisions/assets            | [0.81]       |               | [0.60]       |              | [0.59]        |              | [0.65]      |               |
| Nonperforming                |              | -0.109        |              | -0.143       |               | -0.169*      |             | -0.147        |
| loans/assets                 |              | [1.20]        |              | [1.17]       |               | [1.66]       |             | [1.46]        |
| ln(assets)                   | $0.178^{*}$  | 0.145         | 0.231*       | $0.219^{*}$  | 0.143         | 0.120        | 0.135       | 0.096         |
|                              | [1.96]       | [1.56]        | [1.86]       | [1.74]       | [1.29]        | [1.09]       | [1.38]      | [0.95]        |
| Squared ln(assets)           | $0.145^{*}$  | 0.136         | $0.154^{*}$  | $0.156^{**}$ | $0.152^{**}$  | 0.136*       | 0.113       | 0.094         |
|                              | [1.75]       | [1.60]        | [1.93]       | [1.97]       | [1.99]        | [1.72]       | [1.37]      | [1.10]        |
| Equity/assets                | -0.004*      | -0.004        | $-0.005^{*}$ | $-0.005^{*}$ | $-0.005^{*}$  | $-0.004^{*}$ | -0.003      | -0.003        |
|                              | [1.66]       | [1.57]        | [1.87]       | [1.96]       | [1.91]        | [1.69]       | [1.25]      | [1.03]        |
| Conglomerate affiliated      | 0.039        | 0.026         | 0.054        | 0.059        | 0.060         | 0.080        | $0.066^{*}$ | $0.063^{*}$   |
|                              | [0.57]       | [0.37]        | [0.69]       | [0.75]       | [0.85]        | [1.19]       | [1.88]      | [1.78]        |
| Hyper-regulated period       | 0.001        | 0.003         | -0.006       | -0.002       | -0.001        | 0.003        | -0.005      | -0.002        |

|              | [0.09] | [0.21] | [0.43] | [0.12] | [0.09] | [0.17] | [0.39] | [0.14] |
|--------------|--------|--------|--------|--------|--------|--------|--------|--------|
| Observations | 463    | 463    | 463    | 463    | 463    | 463    | 463    | 463    |
| R-square     | 0.05   | 0.05   | 0.04   | 0.04   | 0.04   | 0.05   | 0.04   | 0.05   |
| Adj R-square | 0.02   | 0.02   | 0.01   | 0.01   | 0.01   | 0.02   | 0.01   | 0.02   |
| F-statistics | 2.39   | 2.19   | 1.66   | 1.74   | 2.13   | 2.15   | 3.32   | 3.51   |

## **Center for Economic Institutions Working Paper Series**

| 2000-1  | Jean Tirole, "Corporate Governance," January 2000.  |
|---------|---|
| 2000-2  | Kenneth A. Kim and S. Ghon Rhee, "A Note on Shareholder Oversight and the Regulatory Environment: The Japanese Banking Experience," January 2000.                                   |
| 2000-3  | S. Ghon Rhee, "Further Reforms after the "BIG BANG": The JapaneseGovernment Bond Market,"June 2000.   |
| 2000-4  | Stijn Claessens, Simeon Djankov , Joseph Fan , and Larry Lang, "Expropriation of Minority Shareholders in East Asia,"July 2000.   |
| 2000-5  | Stijn Claessens, Simeon Djankov, Joseph Fan, and Larry Lang, "The Costs of Group Affiliation: Evidence from East Asia," July 2000.  |
| 2001-1  | Masaharu Hanazaki and Akie Takeuchi, "An International Comparison of Corporate<br>Investment Behavior -Some Implications for the Governance Structure in Japan-," February<br>2001. |
| 2001-2  | Katsuyuki Kubo, "The Determinants of Executive Compensation in Japan and the UK: Agency Hypothesis or Joint Determination Hypothesis?" February 2001.                               |
| 2001-3  | Katsuyuki Kubo, "Changes in Directors' Incentive Plans and the Performance of Firms in the UK," March 2001.   |
| 2001-4  | Yupana Wiwattanakantang, "Controlling Shareholders and Corporate Value: Evidence from Thailand," March 2001.  |
| 2001-5  | Katsuyuki Kubo, "The Effect of Managerial Ownership on Firm Performance: Case in Japan," March 2001.  |
| 2001-6  | Didier Guillot and James R. Lincoln, "The Permeability of Network Boundaries: Strategic Alliances in the Japanese Electronics Industry in the 1990s," March 2001.                   |
| 2001-7  | Naohito Abe, "Ageing and its Macroeconomic Implications-A Case in Japan-," May 2001.  |
| 2001-8  | Yupana Wiwattanakantang, "The Equity Ownership Structure of Thai Firms," July 2001.   |
| 2001-9  | Megumi Suto, "Capital Structure and Investment Behaviour of Malaysian Firms in the 1990sA study of Corporate Governance before the Crisis," August 2001.                            |
| 2001-10 | Naohito Abe, Noel Gaston, and Katsuyuki Kubo, "Executive Pay in Japan : The Role of Bank-Appointed Monitors and the Main Bank Relationship," September 2001.                        |
| 2001-11 | Colin Mayer, "The Financing and Governance of New Technologies," September 2001.  |
| 2001-12 | Masaharu Hanazaki and Akiyoshi Horiuchi, "Can the Financial Restraint Hypothesis Explain Japan's Postwar Experience?" September 2001.   |
| 2001-13 | Shin-ichi Fukuda, "The Role of Long-term Loans for Economic Development: Empirical Evidence in Japan, Korea, and Taiwan," September 2001.   |

| 2001-14 | S. Ghon Rhee, "Further Reforms of the JGB Market for the Promotion of Regional Bond Markets," September 2001.  |
|---------|--|
| 2001-15 | Stijn Claessens, Simeon Djankov, Joseph P. H. Fan, and Larry H. P. Lang, "The Benefits<br>and Costs of Internal Markets: Evidence from Asia's Financial Crisis," September 2001.                               |
| 2001-16 | Kenneth A. Kim and John R. Nofsinger, "Institutional Herding, Business Groups, and Economic Regimes: Evidence from Japan," September 2001.   |
| 2001-17 | Mitsuhiro Fukao, "Financial Deregulations, Weakness of Market Discipline, and Market Development: Japan's Experience and Lessons for Developing Countries," September 2001.                                    |
| 2001-18 | Akio Kuroda and Koichi Hamada, "Towards an Incentive Compatible Financial System:<br>Accounting and Managing the Non-Performing Loans," September 2001.  |
| 2001-19 | Randall Morck and Bernard Yeung, "Japanese Economic Success and the Curious Characteristics of Japanese Stock Prices," September 2001.   |
| 2001-20 | Miguel A. García-Cestona, "Ownership Structure, Banks and the Role of Stakeholders: The Spanish Case," September 2001.   |
| 2001-21 | Joseph P. H. Fan and T. J. Wong, "Corporate Ownership Structure and the Informativeness of Accounting Earnings in East Asia," September 2001.  |
| 2001-22 | Heather Montgomery, "The Effect of the Basel Accord on Bank Lending in Japan," September 2001.   |
| 2001-23 | Naoyuki Yoshino, Sahoko Kaji, and Ayako Suzuki, "The Basket-peg, Dollar-peg and FloatingA Comparative Analysis of Exchange Rate Regimes," September 2001.  |
| 2001-24 | Colin Mayer, Koen Schoors, and Yishay Yafeh, "Sources of Funds and Investment<br>Strategies of Venture Capital Funds: Evidence from Germany, Israel, Japan and the UK,"<br>September 2001.                     |
| 2001-25 | Yukinobu Kitamura, Megumi Suto, and Juro Teranishi, "Towards a New Architecture for<br>the Japanese Financial System: Participation Costs, Intermediated Ownership and Wealth<br>Distribution,"September 2001. |
| 2002-1  | Evgeni Peev, "The Political Economy of Corporate Governance Change in Bulgaria:<br>Washington Consensus, Primitive Accumulation of Capital, and Catching-Up in the 1990,"<br>March 2002.                       |
| 2002-2  | Naohito Abe, "Saving, Capital Flows, and the Symmetric International Spillover of Industrial Policies," June 2002.   |
| 2002-3  | Masaharu Hanazaki and Akiyoshi Horiuchi, "A Review of Japan's Bank Crisis from the Governance Perspective," July 2002.   |
| 2002-4  | Chutathong Charumirind, Raja Kali and Yupana Wiwattanakantang, "Crony Lending:<br>Thailand before the Financial Crisis," September 2002.   |
| 2002-5  | Maitreesh Ghatak and Raja Kali, "Financially Interlinked Business Groups," September 2002.   |
| 2002-6  | Tarun Khanna, Joe Kogan, and Krishna Palepu, "Globalization and Similarities in Corporate Governance: A Cross-Country Analysis," September 2002.   |

| 2002-7  | Chongwoo Choe, "Delegated Contracting and Corporate Hierarchies," September 2002.  |
|---------|--|
| 2002-8  | Tarun Khanna and Yishay Yafeh, "Business Groups and Risk Sharing around the World," September 2002.  |
| 2002-9  | Yitae Kim, Kwangwoo Park, Ronald A. Ratti, and Hyun-Han Shin, "Do Main Banks Extract<br>Rents from their Client Firms? Evidence from Korean Chaebol," September 2002.  |
| 2002-10 | Armen Hovakimian, Edward J. Kane and Luc Laeven, "How Country and Safety-Net Characteristics Affect Bank Risk-Shifting," September 2002.   |
| 2002-11 | Vidhan K. Goyal and Takeshi Yamada, "Asset Price Shocks, Financial Constraint, and Investment: Evidence from Japan," September 2002.   |
| 2002-12 | Clive S. Lennox, "Opinion Shopping and Audit Committees," September 2002.  |
| 2002-13 | Seki Obata, "Pyramid Business Groups in East Asia: Insurance or Tunneling?," September 2002.   |
| 2002-14 | Ishtiaq Pasha Mahmood and Will Mitchell, "Two Faces: Effects of Business Groups on Innovation in Emerging Economies," September 2002.  |
| 2002-15 | Kwangwoo Park, "Foreign Ownership and Firm Value in Japan," September 2002.  |
| 2002-16 | Adrian van Rixtel, Yupana Wiwattanakantang, Toshiyuki Souma, and Kazunori Suzuki,<br>"Banking in Japan: Will "To Big To Fail" Prevail?" December 2002.   |
| 2002-17 | Stijn Claessens and Leora F. Klapper, "Bankruptcy around the World: Explanations of its Relative Use," December 2002.  |
| 2003-1  | Anya Khanthavit, Piruna Polsiri, and Yupana Wiwattanakantang, "Did Families Lose or Gain Control after the East Asian Financial Crisis?" February 2003.  |
| 2003-2  | Hidenobu Okuda, Hidetoshi Hashimoto, and Michiko Murakami, "The Estimation of<br>Stochastic Cost Functions of Malaysian Commercial Banks and Its Policy Implications to<br>Bank Restructuring," February 2003. |
| 2003-3  | Masaharu Hanazaki and Liuqun, "Asian Crisis and Corporate Governance, (in Japanese)" March 2003.   |
| 2003-4  | Fukuju Yamazaki and Hiroyuki Seshita, "Economic Analysis of Bankruptcy law in Japan, (in Japanese)" February 2003.   |
| 2003-5  | Hirofumi Uchida and Hiroshi Osano, "Bank Monitoring and Corporate Governance in Japan, (in Japanese)" March 2003.  |
| 2003-6  | Fukunari Kimura and Kozo Kiyota, "Foreign Ownership and Corporate Performance:<br>Evidence from Japanese Micro Data, (in Japanese)" March 2003.  |
| 2003-7  | Yukinobu Kitamura, "Corporate Profit and Debt- Panel Data Analysis of The Japanese Firms in the 1990s, (in Japanese)" March 2003.  |
| 2003-8  | Chaiyasit Aunchitworawong, Toshiyuki Soma, and Yupana Wiwattanakantang, "Do<br>Families Control Banks Prevail after the East Asia Financial Crisis? Evidence from<br>Thailand" March 2003.                     |

| 2003-9  | Junko Maru, Yasuhiro Yonezawa and Yuki Matsumoto, "Corporate Governance by Foreign<br>Investors in East Asia Corporations (in Japanese)" March 2003.                                       |
|---------|--|
| 2003-10 | Sui Qing-yuan, "Declining Firm's Dependence upon Bank Borrowing and Corporate Performance (in Japanese)" March 2003.   |
| 2003-11 | Katsumi Matsuura, "Changes in Ownership Structures and Their Impacts upon Corporate Performance in Japan (in Japanese)" March 2003.  |
| 2003-12 | Kathy S. He, Randall Morck and Bernard Yeung, "Corporate Stability and Economic Growth," May 2003.   |
| 2003-13 | Robert Dekle and Heajin Ryoo, "Exchange Rate Fluctuations, Financing Constraints, Hedging, and Exports: Evidence from Firm Level Data," June 2003.   |
| 2003-14 | Tsun-Siou Lee, Yin-Hua Yeh and Rong-Tze Liu, "Can Corporate Governance Variables<br>Enhance the Prediction Power of Accounting-Based Financial Distress Prediction Models?,"<br>June 2003. |
| 2003-15 | Hideaki Miyajima and Yishay Yafeh, "Japan's Banking Crisis: Who has the Most to Lose?," June 2003.   |
| 2003-16 | Guifen Pei, "Asset Management Companies in China," June 2003.  |
| 2003-17 | Takeshi Nagase, "The Governance Structure of IPO Firm in Japan," July 2003.  |
| 2003-18 | Masaharu Hanazaki and Qun Liu, "The Asian Crisis and Corporate Governance — Ownership Structure, Debt Financing, and Corporate Diversification — ," July 2003.                             |
| 2003-19 | Chutatong Charumilind, Raja Kali and Yupana Wiwattanakantang, "Connected Lending:<br>Thailand before the Financial Crisis," July 2003.   |
| 2003-20 | Gilles Hilary and Tomoki Oshika, "Shareholder activism in Japan: social pressure, private cost and organized crime," August 2003.  |
| 2003-21 | Sanghoon Ahn, "Technology Upgrading with Learning Cost," September 2003.   |
| 2003-22 | Masaharu Hanazaki and Akiyoshi Horiuchi, "Have Banks Contributed to Efficient Management in Japan's Manufacturing?," November 2003.  |
| 2003-23 | Chongwoo Choe and In-Uck Park, "Delegated Contracting and Corporate Hierarchies," November 2003.   |
| 2003-24 | Bruno Dallago, "Comparative Economic Systems and the New Comparative Economics: Foes, Competitors, or Complementary?," November 2003.  |
| 2003-25 | Adrian van Rixtel, Ioana Alexopoulou and Kimie Harada, "The New Basel Capital Accord<br>and Its Impact on Japanese Banking: A Qualitative Analysis," November 2003.                        |
| 2004-1  | Masaharu Hanazaki, Toshiyuki Souma and Yupana Wiwattanakantang, "Silent Large<br>Shareholders and Entrenched Bank Management: Evidence from Banking Crisis in Japan,"<br>January 2004.     |
| 2004-2  | Ming Ming Chiu and Sung Wook Joh, "Bank Loans to Distressed Firms: Cronyism, bank governance and economic crisis," January 2004.   |

| 2004-3  | Keun Lee, Keunkwan Ryu and Jungmo Yoon, "Corporate Governance and Long Term Performance of the Business Groups: The Case of Chaebols in Korea," January 2004.                               |
|---------|---|
| 2004-4  | Randall Morck and Masao Nakamura, "Been There, Done That – The History of Corporate Ownership in Japan," March 2004.  |
| 2004-5  | Dong-Hua Chen, Joseph P. H. Fan and T. J. Wong, "Politically-connected CEOs, Corporate Governance and Post-IPO Performance of China's Partially Privatized Firms," March 2004.              |
| 2004-6  | Jae-Seung Baek, Jun-Koo Kang and Inmoo Lee, "Business Groups and Tunneling: Evidence from Private Securities Offerings by Korean Chaebols," March 2004.                                     |
| 2004-7  | E. Han Kim, "To Steal or Not to Steal: Firm Attributes, Legal Environment, and Valuation," March 2004.  |
| 2004-8  | Yin-Hua Yeh and Tracie Woidtke, "Commitment or Entrenchment?: Controlling Shareholders and Board Composition," June 2004.   |
| 2004-9  | Hugh Patrick, "Thoughts on Evolving Corporate Governance in Japan," June 2004.  |
| 2004-10 | Utpal Bhattacharya and Hazem Daouk, "When No Law is Better than a Good Law", June 2004.   |
| 2004-11 | Sanghoon Ahn, Utpal Bhattacharya, Taehun Jung and Giseok Nam, "Do Japanese CEOs Matter?", June 2004.  |
| 2004-12 | Megumi Suto and Masashi Toshino, "Behavioural Biases of Japanese Institutional<br>Investors; Fund management and Corporate Governance", July 2004.  |
| 2004-13 | Piruna Polsiri and Yupana Wiwattanakantang, "Business Groups in Thailand: Before and after the East Asian Financial Crisis", August 2004.   |
| 2004-14 | Fumiharu Mieno, "Fund Mobilization and Investment Behavior in Thai Manufacturing Firms in the Early 1990s", August 2004.  |
| 2004-15 | Chaiyasit Anuchitworawong, "Deposit Insurance, Corporate Governance and Discretionary Behavior: Evidence from Thai Financial Institutions", September 2004.                                 |
| 2004-16 | Chaiyasit Anuchitworawong, "Financial fragility under implicit insurance scheme: Evidence from the collapse of Thai financial institutions", September 2004.                                |
| 2004-17 | Chaiyasit Anuchitworawong, "Ownership-based Incentives, Internal Corporate Risk and Firm Performance", September 2004.  |
| 2004-18 | Jack Ochs and In-Uck Park, "Overcoming the Coordination Problem: Dynamic Formation of Networks", September 2004.  |
| 2004-19 | Hidenobu Okuda and Suvadee Rungsomboon, "Comparative Cost Study of Foreign and<br>Thai Domestic Banks 1990–2002: Estimating Cost Functions of the Thai Banking<br>Industry," February 2005. |
| 2004-20 | Hidenobu Okuda and Suvadee Rungsomboon, "The Effects of Foreign Bank Entry on the Thai Banking Market: Empirical Analysis from 1990 to 2002," March 2005.                                   |

| 2004-21 | Juro Teranishi, "Investor Right in Historical Perspective: Globalization and the Future of the Japanese Firm and Financial System," March 2005.                        |
|---------|--|
| 2004-22 | Kentaro Iwatsubo, "Which Accounts for Real Exchange Rate Fluctuations, Deviations from the Law of One Price or Relative Price of Nontraded Goods?", March 2005.        |
| 2004-23 | Kentaro Iwatsubo and Tomoyuki Ohta, "Causes and effects of exchange rate regimes (in Japanese)," March 2005.   |
| 2004-24 | Kentaro Iwatsubo, "Bank Capital Shocks and Portfolio Risk: Evidence from Japan," March 2005.   |
| 2004-25 | Kentaro Iwatsubo, "On the Bank-led Rescues Financially Distressed Firms in Japan," March 2005.   |
| 2005-1  | Yishay P. Yafeh and Tarun Khanna, "Business Groups in Emerging Markets: Paragons or Parasities?," September 2005.  |
| 2005-2  | Renee B. Adams and Daniel Ferreira, "Do Directors Perform for Pay?," September 2005.   |
| 2005-3  | Qun Liu, Shin-ichi Fukuda and Juro Teranishi, "What are Characteristics of Financial Systems in East Asia as a Region?." September 2005.                               |
| 2005-4  | Juro Teranishi, "Is the Financial System of Postwar Japan Bank-dominated or Market Based?," September 2005.  |
| 2005-5  | Hasung Jang, Hyung-cheol Kang and Kyung Suh Park, "Determinants of Family Ownership: The Choice between Control and Performance," October 2005.                        |
| 2005-6  | Hasung Jang, Hyung-cheol Kang and Kyung Suh Park, "The Choice of Group Structure: Divide and Rule," October 2005.  |
| 2005-7  | Sangwoo Lee, Kwangwoo Park and Hyun-Han Shin, "The Very Dark Side of International Capital Markets: Evidence from Diversified Business Groups in Korea," October 2005. |
| 2005-8  | Allen N. Berger, Richard J. Rosen and Gregory F. Udell, "Does Market Size Structure Affect Competition? The Case of Small Business Lending," November 2005.            |
| 2005-9  | Aditya Kaul and Stephen Sapp, "Trading Activity and Foreign Exchange Market Quality," November 2005.   |
| 2005-10 | Xin Chang, Sudipto Dasgupta and Gilles Hilary, "The Effect of Auditor Choice on Financing Decisions," December 2005.   |
| 2005-11 | Kentaro Iwatsubo, "Adjustment Speeds of Nominal Exchange Rates and Prices toward<br>Purchasing Power Parity," January 2006.  |
| 2005-12 | Giovanni Barone-Adesi, Robert Engle and Loriano Mancini, "GARCH Options in Incomplete Markets", March 2006.  |
| 2005-13 | Aditya Kaul, Vikas Mehrotra and Blake Phillips, "Ownership, Foreign Listings, and Market Valuation", March 2006.   |
| 2005-14 | Ricard Gil, "Renegotiation, Learning and Relational Contracting", March 2006.  |

| 2005-15 | Randall Morck, "How to Eliminate Pyramidal Business Groups -The Double Taxation of<br>Inter-corporate Dividends and other Incisive Uses of Tax Policy-", March 2006.                                   |
|---------|--|
| 2005-16 | Joseph P.H. Fan, T.J. Wong and Tianyu Zhang, "The Emergence of Corporate Pyramids in China", March 2006.   |
| 2005-17 | Yan Du, Qianqiu Liu and S. Ghon Rhee, "An Anatomy of the Magnet Effect: Evidence from the Korea Stock Exchange High-Frequency Data", March 2006.   |
| 2005-18 | Kentaro Iwatsubo and Junko Shimizu, "Signaling Effects of Foreign Exchange Interventions and Expectation Heterogeneity among Traders", March 2006.   |
| 2005-19 | Kentaro Iwatsubo, "Current Account Adjustment and Exchange Rate Pass-Through(in Japanese)", March 2006.  |
| 2005-20 | Piruna Polsiri and Yupana Wiwattanakantang, "Corporate Governance of Banks in Thailand", March 2006.   |
| 2006-1  | Hiroyuki Okamuro and Jian Xiong Zhang, "Ownership Structure and R&D Investment of Japanese Start-up Firms," June 2006.   |
| 2006-2  | Hiroyuki Okamuro, "Determinants of R&D Activities by Start-up Firms: Evidence from Japan," June 2006.  |
| 2006-3  | Joseph P.H. Fan, T.J. Wong and Tianyu Zhang, "The Emergence of Corporate Pyramids in China," August 2006.  |
| 2006-4  | Pramuan Bunkanwanicha, Jyoti Gupta and Yupana Wiwattanakantang, "Pyramiding of Family-owned Banks in Emerging Markets," September 2006.  |
| 2006-5  | Bernardo Bortolotti and Mara Faccio, "Reluctant privatization," September 2006.  |
| 2006-6  | Jörn Kleinert and Farid Toubal, "Distance costs and Multinationals' foreign activities", October 2006.   |
| 2006-7  | Jörn Kleinert and Farid Toubal, "Dissecting FDI", October 2006.  |
| 2006-8  | Shin-ichi Fukuda and Satoshi Koibuchi, "The Impacts of "Shock Therapy" on Large and Small Clients: Experiences from Two Large Bank Failures in Japan", October 2006.                                   |
| 2006-9  | Shin-ichi Fukuda, Munehisa Kasuya and Kentaro Akashi, "The Role of Trade Credit for Small Firms: An Implication from Japan's Banking Crisis", October 2006.  |
| 2006-10 | Pramuan Bunkanwanicha and Yupana Wiwattanakantang, "Big Business Owners and<br>Politics: Investigating the Economic Incentives of Holding Top Office", October 2006.                                   |
| 2006-11 | Sang Whi Lee, Seung-Woog(Austin) Kwang, Donald J. Mullineaux and Kwangwoo Park,<br>"Agency Conflicts, Financial Distress, and Syndicate Structure: Evidence from Japanese<br>Borrowers", October 2006. |
| 2006-12 | Masaharu Hanazaki and Qun Liu, "Corporate Governance and Investment in East Asian Firms -Empirical Analysis of Family-Controlled Firms", October 2006.   |
| 2006-13 | Kentaro Iwatsubo and Konomi Tonogi, "Foreign Ownership and Firm Value: Identification through Heteroskedasticity (in Japanese)", December 2006.  |

| 2006-14 | Kentaro Iwatsubo and Kazuyuki Inagaki, "Measuring Financial Market Contagion Using Dually-Traded Stocks of Asian Firms", December 2006.   |
|---------|---|
| 2006-15 | Hun-Chang Lee, "When and how did Japan catch up with Korea? –A comparative study of the pre-industrial economies of Korea and Japan", February 2007.  |
| 2006-16 | Kyoji Fukao, Keiko Ito, Shigesaburo Kabe, Deqiang Liu and Fumihide Takeuchi, "Are Japanese Firms Failing to Catch up in Localization? An Empirical Analysis Based on Affiliate-level Data of Japanese Firms and a Case Study of the Automobile Industry in China", February 2007. |
| 2006-17 | Kyoji Fukao, Young Gak Kim and Hyeog Ug Kwon, "Plant Turnover and TFP Dynamics in Japanese Manufacturing", February 2007.   |
| 2006-18 | Kyoji Fukao, Keiko Ito, Hyeg Ug Kwon and Miho Takizawa, "Cross-Border Acquisitons and Target Firms' Performance: Evidence from Japanese Firm-Level Data", February 2007.  |
| 2006-19 | Jordan Siegel and Felix Oberholzer-Gee, "Expropriators or Turnaround Artists? The Role of Controlling Families in South Korea (1985-2003)", March 2007.   |
| 2006-20 | Francis Kramarz and David Thesmar, "Social Networks in The Boardroom", March 2007.  |
| 2006-21 | Morten Bennedsen, Francisco Pérez-González and Daniel Wolfenzon, "Do CEOs matter?",<br>March 2007.  |
| 2007-1  | Ichiro Iwasaki, "Endogenous board formation and its determinants in a transition economy: evidence from Russia*", April 2007, Revised on October 2007.  |
| 2007-2  | Joji Tokui, Tomohiko Inui, and Katsuaki Ochiai, "The Impact of Vintage Capital and R&D on Japanese Firms' Productivity", April 2007.  |
| 2007-3  | Yasuo Nakanishi and Tomohiko Inui, "Deregulation and Productivity in Japanese Industries", April 2007.  |
| 2007-4  | Kyoji Fukao, "The Performance of Foreign Firms and the Macroeconomic Impact of FDI", May 2007.  |
| 2007-5  | Taku Suzuki, "The Role of the State in Economic Growth of Post-Communist Transitional Countries", June 2007.  |
| 2007-6  | Michiel van Leuvensteijn, Jacob A. Bikker, Adrian A.R.J.M. van Rixtel and Christoffer Kok-Sørensen*, "A new approach to measuring competition in the loan markets of the euro area", June 2007.   |
| 2007-7  | Sea Jin Chang, Jaiho Chung, and Dean Xu, "FDI and Technology Spillovers in China", July 2007.   |
| 2007-8  | Fukunari Kimura, "The mechanics of production networks in Southeast Asia: the fragmentation theory approach", July 2007.  |
| 2007-9  | Kyoji Fukao, Tsutomu Miyagawa, Miho Takizawa, "Productivity Growth and Resource Reallocation in Japan", November 2007.  |
| 2007-10 | YoungGak Kim, "A Survey on Intangible Capital", December 2007.  |

| 2007-11 | Sea-Jing Chang and Jay Hyuk Rhee, "Rapid International Expansion Strategy of Emerging<br>Market Enterprises: The Interplay between Speed and Competitive Risks on International<br>performance", November 2007. |
|---------|---|
| 2007-12 | Ishtiaq Mahmood, Will Mitchell, and Chi-Nien Chung, "The Structure of Intra-Group Ties:<br>Innovation in Taiwanese Business", January 2008.   |
| 2007-13 | Kyoji Fukao, Tomohiko Inui, Shigesaburo Kabe and Deqiang Liu, "An International Comparison of the TFP Levels of Japanese, Korean and Chinese Listed Firms", March 2008.   |
| 2007-14 | Pramuan Bunkanwanicha and Yupana Wiwattanakantang, "Allocating Risk Across<br>Pyramidal Tiers: Evidence from Thai Business Groups", March 2008.   |
| 2008-1  | Rüdiger Fahlenbrach and René M. Stulz, "Managerial Ownership Dynamics and Firm Value", April 2008.  |
| 2008-2  | Morten Bennedsen, Kasper Meisner Nielsen, and, Thomas Vester Nielsen, "Private<br>Contracting and Corporate Governance: Evidence from the Provision of Tag-Along Rights<br>in an Emerging Market", April 2008.  |
| 2008-3  | Joseph P.H. Fan, Jun Huang, Felix Oberholzer-Gee, and Mengxin Zhao, "Corporate Diversification in China: Causes and Consequences", April 2008.  |
| 2008-4  | Daniel Ferreira, Miguel A. Ferreira, Clara C. Raposo, "Board Structure and Price Informativeness", April 2008.  |
| 2008-5  | Nicola Gennaioli and Stefano Rossi, "Judicial Discretion in Corporate Bankruptcy", April 2008.  |
| 2008-6  | Nicola Gennaioli and Stefano Rossi, "Optimal Resolutions of Financial Distress by Contract", April 2008.  |
| 2008-7  | Renée B. Adams and Daniel Ferreira, "Women in the Boardroom and Their Impact on Governance and Performance", April 2008.  |
| 2008-8  | Worawat Margsiri, Antonio S. Melloy, and Martin E. Ruckesz, "A Dynamic Analysis of Growth via Acquisition", April 2008.   |
| 2008-9  | Pantisa Pavabutra and Sukanya Prangwattananon, "Tick Size Change on the Stock Exchange of Thailand", April 2008.  |
| 2008-10 | Maria Boutchkova, Hitesh Doshi, Art Durnev, and Alexander Molchanov, "Politics and Volatility", April 2008.   |
| 2008-11 | Yan-Leung Cheung, P. Raghavendra Rau, and Aris Stouraitis, "The Helping Hand, the Lazy Hand, or the Grabbing Hand? Central vs. Local Government Shareholders in Publicly Listed Firms in China", April 2008.    |
| 2008-12 | Art Durnev and Larry Fauver, "Stealing from Thieves: Firm Governance and Performance when States are Predatory", April 2008.  |
| 2008-13 | Kenneth Lehn, Sukesh Patro, and Mengxin Zhao, "Determinants of the Size and Structure of Corporate Boards: 1935-2000", April 2008.  |

| 2008-14 | Ishtiaq P. Mahmood, Hong-Jin Zhu and Edward J. Zajac, "Where Can Capabilities Come<br>From? How the Content of Network Ties Affects Capability Acquisition", April 2008.   |
|---------|--|
| 2008-15 | Vladimir I. Ivanov and Ronald W. Masulis, "Corporate Venture Capital, Strategic Alliances, and the Governance of Newly Public Firms", May 2008.  |
| 2008-16 | Dick Beason, Ken Gordon, Vikas Mehrotra and Akiko Watanabe, "Does Restructuring Pay in Japan? Evidence Following the Lost Decade", July 2008 (revision uploaded on Oct. 2009).                                       |
| 2009-1  | Vikas Mehrotra, Dimitri van Schaik, Jaap Spronk, and Onno Steenbeek, "Creditor-Focused Corporate Governance: Evidence from Mergers and Acquisitions in Japan," August, 2009.   |
| 2009-2  | Debin Ma, "Law and Economic Change in Traditional China: A Comparative Perspective," September, 2009.  |
| 2009-3  | Robert C. Allen, Jean-Pascal Bassino, Debin Ma, Christine Moll-Murata, and Jan Luiten van Zanden, "Wages, Prices, and Living Standards in China, 1738-1925: in Comparison with Europe, Japan, and India," June 2009. |
| 2009-4  | Jung-Wook Shim, "The Existence of Nepotism: Evidence from Japanese Family Firms,"<br>October 2009.   |
| 2009-5  | Morten Bennedsen and Kasper Meisner Nielsen, "Incentive and Entrenchment Effects in European Ownership," March 2009.   |
| 2009-6  | Joseph P.H. Fan, TJ Wong, Tianyu Zhang, "Founder Succession and Accounting Properties,"<br>April 2009.   |
| 2009-7  | Hiroyuki Okamuro, Masatoshi Kato, and Yuji Honjo, "Determinants of R&D Cooperation in Japanese High-tech Start-ups," November 2009.  |
| 2009-8  | Bill Francis, Iftekhar Hasan, Michael Koetter, and Qiang Wu, "The Effectiveness of Corporate Boards: Evidence from Bank Loan Contracting," November 2009.  |
| 2009-9  | Allen N. Berger, Iftekhar Hasan and Mingming Zhou, "The Effects of Focus Versus Diversification on Bank Performance: Evidence from Chinese Banks," November 2009.  |